Analyzing the risk of Maladaptation: Implications for adaptation projects among development agencies

Bikram Sedhai | Division of Risk Management and Societal Safety | LTH | LUND UNIVERSITY, SWEDEN



Analyzing the risk of maladaptation: Implications for adaptation projects among development agencies

Bikram Sedhai

Lund 2023

Analyzing the risk of maladaptation: Implications for adaptation projects among development agencies

Bikram Sedhai

Number of pages: 62 Illustrations: 5

Keywords

Climate Change Adaptation, Maladaptation, Non-Governmental Organizations, Transformational Adaptation, International aid

Abstract

The need for climate change adaptation is widely acknowledged and adaptation projects around the world are proliferating. Despite being one of the key actors for the implementation of adaptation projects, significant knowledge gaps persist regarding the potential maladaptive outcomes of such projects for international non-governmental organizations. This study analyses practitioners' perspectives on effective adaptation, maladaptation, and existing challenges and opportunities to provide recommendations for better practice. A qualitative descriptive case study was conducted using a mixed-methods approach, including 15 semi-structured interviews, a systematic literature review, and project document analysis. The research highlights the need for a holistic understanding of successful adaptation, with justice and equity as the central focus of adaptation planning. This study also identifies key challenges in addressing the risk of maladaptation, including funding mechanisms, insufficient contextual understanding, conceptual ambiguity, inadequate adaptive learning, and poor project practice. Additionally, this paper contributes to the emerging literature on maladaptation from practitioners' perspectives and provides recommendations and guiding questions to consider potential adverse outcomes of projects. Ultimately, this research demonstrates that maladaptation could be a useful tool to transform adaptation practices among organizations and ensure sustainable and effective adaptation projects.

© Copyright: Division of Risk Management and Societal Safety, Faculty of Engineering Lund University, Lund 2023 Avdelningen för Riskhantering och samhällssäkerhet, Lunds tekniska högskola, Lunds universitet, Lund 2023.

> Riskhantering och samhällssäkerhet Lunds tekniska högskola Lunds universitet Box 118 221 00 Lund

> > http://www.risk.lth.se

Telefon: 046 - 222 73 60

Division of Risk Management and Societal Safety Faculty of Engineering Lund University P.O. Box 118 SE-221 00 Lund Sweden

http://www.risk.lth.se

Telephone: +46 46 222 73 60

Acknowledgements

Firstly, I would like to express my sincere gratitude to everyone who has been part of the thesis, both directly and indirectly. In particular, I would like to thank all the interviewees and organizational representatives for the invaluable inputs and willingness to share insights, this thesis would not have been possible without your contribution (Please note that organizations have not been mentioned by name, as agreed prior to the research commencing).

I am deeply grateful to my supervisor, Mo Hamza, for his continuous support and feedback throughout the research process. Thank you for providing thoughtful supervision whenever needed while also allowing me to work at my own pace. I would also like to extend my thanks to the team at Barnfonden, where I undertook my internship, for listening to my research idea since the very beginning. Thank you for all those encouraging discussions and talks and providing generous help throughout this research process.

I am thankful to Martin Rokitzi, Humayain Kabir, and Lingaraj Jayaprakash from PlanAdapt for the thought-provoking ideas and suggestions, which has shaped my further research interest in this topic.

Special thanks to all the teachers at the Division of Risk Management and Societal Safety and my classmates for the enriching learning journey. Finally, and with utmost importance, I would like to thank the Swedish Institute Scholarships for Global Professionals for providing me with the scholarship to pursue my master's degree.

Once again, thank you to all who have contributed to this thesis. Your support and encouragement have been greatly appreciated.

Summary

In recent years, the urgency to adapt to climate change has gained significant attention, as extreme weather events have become more frequent and intense due to anthropogenic activities. Bilateral or multilateral aid is anticipated to increase for adaptation projects in vulnerable countries, which are often implemented through development agencies. Thus, it is crucial to analyze adaptation practices among these organizations and provide recommendations for improved planning and implementation. Similarly, critical adaptation literature has raised concerns about maladaptation, along with increasing evidence of adaptation failure. However, the operationalization of this concept is not implemented among the principal implementers of adaptation projects. Therefore, this study aimed to address these conceptual and practical gaps, identify potential ways to address the risk of maladaptation, and raise awareness among development practitioners regarding rethinking adaptation practices.

The research was conducted using a primarily inductive approach through a qualitative descriptive study. Primary data were collected through 15 semi-structured interviews, and secondary data were collected through a review of academic and gray literature, as well as three projects in India, Ethiopia, and Zambia. Chapter 3 introduces the key foundational concepts that guided the research, and Chapter 4 provides a comprehensive overview of mainstream adaptation practices, emerging typologies of maladaptation, and existing frameworks and guidelines. Chapter 5 briefly introduces the projects and the key components included in the case study. In Chapter 6, the findings from the interviews and document reviews are presented and discussed in comparison to academic and grey literature. The findings indicate different interrelated factors in adaptation practice, which were categorized into eight main themes: (1) understanding contextual factors; (2) justice and equity; (3) inadequate monitoring and evaluation; (4) finance structure and mechanism; (5) conceptual ambiguity; (6) adaptive learning culture; (7) shifting towards a holistic perspective; and (8) poor project practice. Chapter 7 summarizes the key findings and provides recommendations and guiding questions to reflect on.

The study found that effective adaptation is characterized by a locally led, community-based approach, with a shared understanding of justice and equity at the center of adaptation planning. Overlooking contextual factors and limited monitoring and evaluation were identified as key determinants of maladaptive outcomes. Therefore, organizations should incorporate comprehensive participatory and contextually driven analyses and co-create project tools with the intended beneficiaries of the projects. Similarly, structural issues in funding mechanisms, conceptual ambiguity, absence of a learning culture, narrow perspective, and poor project practice were identified as key challenges. It is recommended that aid agencies promote patient, predictable, and flexible funding; incorporate a learning culture into future project planning; and promote long-term flexible programming. Lastly, the study suggests that further research should explore maladaptation as a tool to reshape current adaptation practices among organizations, ensuring effective and sustainable adaptation.

Abbreviations

CCA	Climate Change Adaptation
FPCs	Farmer Producer Companies
I/NGOs	International/National Non-Governmental Organisations
IPCC	Intergovernmental Panel on Climate Change
LIAP	Luangwa Integrated Agriculture Project
MEL	Monitoring Evaluation and Learning
M&E	Monitoring and Evaluation
ToC	Theory of Change
UNDRR	United Nations Office for Disaster Risk Reduction

Table of Contents

	8
1.1 Background	8
1.2 Purpose	9
1.3 Research questions	9
1.4 Research Limitations	9
2. Methodology	
2.1 Secondary data collection	
2.2 Primary data collection	
2.3 Data Analysis	
3. Conceptual background	13
3.1 Vulnerability	
3.2 Climate change risk	14
3.3 Adaptation	
3.4 Maladaptation	
3.5 I/NGOs role in climate adaptation	
4. Literature review	
4.1 Adaptation framing	
4.2 Maladaptation and its typologies	20
4.5 Existing frameworks and tools on understanding maladaptation	
4.3 Existing frameworks and tools on understanding maladaptation 5. Case study descriptions	24
 4.5 Existing frameworks and tools on understanding maladaptation 5. Case study descriptions	24
 4.5 Existing frameworks and tools on understanding maladaptation 5. Case study descriptions 5.1 Right-based project with an agricultural component, Ethiopia 5.2 Gender empowerment project supporting climate smart agriculture and economic develop 	
 4.3 Existing frameworks and tools on understanding maladaptation	
 4.3 Existing frameworks and tools on understanding maladaptation	24 24 24 25 25 25
 4.3 Existing frameworks and tools on understanding maladaptation	
 4.3 Existing frameworks and tools on understanding maladaptation	24 24 24 25 25 25 26 26 29
 4.3 Existing frameworks and tools on understanding maladaptation	
 4.3 Existing frameworks and tools on understanding maladaptation	24 24 24 25 25 25 26 26 29 30 30
 4.3 Existing frameworks and tools on understanding maladaptation	
 4.3 Existing frameworks and tools on understanding maladaptation	24
 4.3 Existing frameworks and tools on understanding maladaptation	24 24 24 25 25 25 25 26 29 30 30 32 34 34 35 36
4.3 Existing frameworks and tools on understanding maradaptation	24
4.3 Existing frameworks and tools on understanding maladaptation	24 24 24 25 25 25 25 26 29 30 30 32 34 35 36 38 39
4.3 Existing frameworks and tools on understanding maladaptation	24
4.3 Existing frameworks and tools on understanding manadaptation 5. Case study descriptions 5.1 Right-based project with an agricultural component, Ethiopia 5.2 Gender empowerment project supporting climate smart agriculture and economic develop 5.3 An integrated agricultural project with a larger water component, Zambia 6. Results and Discussion 6.1 Understanding context 6.2 Equity and Justice 6.3 Inadequate monitoring and evaluation 6.4 Funding structure and mechanisms 6.5 Conceptual ambiguity 6.6 Culture of learning 6.7 Shift from shallow understanding towards holistic perspective. 6.8 Project problems 7. Conclusion and Recommendations 8 References 9. Appendices	24 24 24 25 25 25 26 29 30 30 32 34 34 35 36 38 39 45 57
4.3 Existing frameworks and tools on understanding maladaptation	24
4.3 Existing frameworks and tools on understanding maladaptation	24 oment in India
4.3 Existing frameworks and tools on understanding manadaptation	24 24 24 25 25 25 26 29 30 32 34 35 36 38 39 45 57 57 58 59

1. Introduction

1.1 Background

Anthropogenically driven climate change is substantive and there is a dire need to adapt to its adverse effects. There is growing concern and consensus, from local to global level, regarding the urgency to respond to socio-ecological threats posed by climate change (Berrang-Ford et al., 2021; IPCC, 2022). For years, adaptation efforts were being downplayed as they were perceived to be contrary to climate change mitigation; however, this has changed and the rationale for undertaking adaptation are well established (Glover & Granberg, 2020). Until recently, political and scientific debates predominantly focused on mitigation; however, the failure to limit greenhouse gas emissions and address the drivers of climate change has made adaptation a global priority (Bertana et al., 2022; Inderberg et al., 2014).

In this globalized world, the economic, social, and biophysical connections between people and places are becoming more complex (Atteridge & Remling, 2018). This, in turn, leads to concerns about the distributional impact of our adaptation actions (Jones et al., 2015). While academic literature shows increasing awareness of the shift or redistribution of vulnerability (Schipper, 2020), this has not significantly shaped real-world choices by policy makers, project planners, or international funds (Atteridge & Remling, 2018). Similarly, the types of responses that strengthen the adaptive ability of socio-ecological systems are not clearly understood (Magnan et al., 2016). Even worse, there is high possibility that current initiatives might turn out to be harmful in the long term, which is referred to as 'maladaptive outcome' (Magnan, 2014). Despite the ongoing concern that adaptation might end up redistributing risk and vulnerability between people and places, very limited attention has been paid to its meaningful translation into practice (Eriksen et al., 2021). Once the global average temperature rises further by 1.5 degree Celsius, some adaptation efforts will be useless, while others will be detrimental (Schipper, 2022). These can undermine opportunities for adaptation in the future, creating pathways of dependencies (Swatuk et al., 2021). However, scientific discussions on adaptation scholarship so far have not been able to substantially contribute to effective planning and wider discussions on how the adaptation process should be governed (Atteridge & Remling, 2018). Given the traction on adaptation efforts, significant gaps in the knowledge and evaluation of adaptation effectiveness persist (Adger et al., 2009; Boutroue et al., 2022; Singh et al., 2022). Numerous outcomes and pathways may emerge from a wide range of adaptation efforts in practice. While it is important to emphasize adaptation outcomes, it is equally essential to understand the processes and factors influencing them (Bertana et al., 2022). This eventually produces knowledge and helps to improve adaptation efforts. Most studies on maladaptation so far have been limited to case studies in specific contexts and actions, with very little contribution to wider practice. In other words, maladaptation remains a theoretically underdeveloped concept.

Likewise, much of the adaptation research is significantly focused on establishing opportunities and barriers, and much less on implementation. To date, there has been very limited independent review of the impacts of international or bilateral-funded interventions on social vulnerability. Nevertheless, theoretical advancement and reported empirical cases of maladaptation have laid the foundation for examining different aspects of adaptation interventions, including framing, planning, implementation, and evaluation (Chi et al., 2020; Eriksen et al., 2011, 2021). So, identifying the mechanisms through which negative effects can unfold in adaptation interventions is essential for informing future adaptation policy and actions (Jones et al., 2015). International non-governmental organizations are currently a principal implementer of adaptation projects, as climate adaptation has folded into development practice and scholarship (Webber, 2016). International climate funds, development banks, and bilateral development organizations are leading adaptation projects in various parts of the world (Atteridge & Remling, 2018). Therefore, it is important to understand practitioners' perspective on effective adaptation, existing practices to better understand what maladaptation means in practice, and the current debate on maladaptation should continue.

1.2 Purpose

This study aims to examine practitioners' perspective on adaptation and how certain actions might lead to adverse impacts, so that they could be reduced in future planning and implementation of adaptation projects among international non-governmental organizations. The overall objective of the research project is to identify ways of addressing the risk of maladaptation by providing inputs for planning and implementing adaptation projects for development agencies. And most importantly, this research aims to contribute and take the discussion further and raise concern among development practitioners to rethink adaptation practice.

1.3 Research questions

- What are the principles that characterize effective adaptation?
- What are the pre-determining factors that might lead to the negative consequences of adaptation projects?
- What are the challenges in the existing practice of planning and implementing adaptation projects for I/NGOs? How can these be addressed?

1.4 Research Limitations

First, the sample size was relatively small, which may limit the generalizability of the findings to a broad range of organizations involved in climate change adaptation. Although efforts were made to ensure a diverse and representative sample, the results should be interpreted with caution as the initial contact with potential interviewees and the projects were sourced through an NGO network who have a child-focused mandate. It should also be noted that the contexts of the projects have been influenced by the organizational mandate and is evident in some of the project emphasis. Second, the study used a limited number of project documents from which to draw information. This included project design documents, baselines and endlines for each of the three projects. This means some activities may have been conducted, but not made clear in these documents. Third, the study used three available project documents and

information; however, the research did not attempt to label project interventions as adaptive or maladaptive. Instead, the analysis focuses on analyzing current practices, existing tools, and other contents of project documents without a specific categorization. Fourth, while 'maladaptation' has been studied in the literature for about two decades, there is still no consensus in practice, leading to ambiguity and subjectivity in its use. This could potentially affect the interpretation of results. To address this limitation, this study used a widely accepted definition, identified existing conceptualization, and most importantly, acknowledged the interviewee's understanding of the concept. Finally, the researcher's academic background and interest in non-governmental organizations may have introduced bias in the research design, data collection, and analysis. These limitations should be considered when interpreting the results of this study and drawing conclusions.

2. Methodology

This study applies a qualitative descriptive case study approach, utilizing a range of data sources, including project documents, systematic scoping literature review, and semi structured interviews. Considering the complexity and limitations as mentioned above, such a mixed method approach allows the research to get more comprehensive and nuanced understanding of the topic (Yin, 2009). Following the usual characteristics of case studies (Creswell, 2013), research included different sources of primary and secondary data.

2.1 Secondary data collection

Secondary data sources encompassing both scholarly and grey literature were used in this study. SCOPUS was the main database for the peer-reviewed literature using keyword search (figure 1). In addition, other databases, such as LUBsearch and ResearchGate, were used for snowballing from the reference list and accessing some key restricted articles. This ensured, to a great extent, that papers explaining the adverse outcomes and challenges of climate change adaptation without using the keyword 'maladaptation' (since it is a relatively young concept) were not missed.



Figure 1: procedure for the literature review

Regarding the grey literatures, major climate adaptation knowledge portals such as Global Center on Adaptation (GCA) (https://adaptationportal.gca.org), World Resources Institute (WRI) (https://www.wri.org), Adaptation Research Alliance (https://www.adaptationresearchalliance.org) were screened for potential guidelines/frameworks. In addition, project documents and resources of major organizations working in the case area context were accessed and used to validate the existing practices and identify potential themes for analysis. While all the project documents and resources were not analysed in great depth, they were used to provide brief overview on adaptation practice among development agencies.

2.2 Primary data collection

A significant source of the data was semi structured interview with stakeholders and representative involved in various stages of project management. The sampling was based on purposive sampling and snowball sampling depending on potential participants identified through communication with interviewees and emergent knowledge gaps (Flick, 2018). The focus of the interviews was on practitioners involved in internationally funded projects with CCA component. To initiate the process, potential interviewees were identified through the researchers' connections, and invitations to participate in the research interviews were sent via

email to practitioners associated with the case studies. Initially, seven practitioners and experts were contacted. However, due to demanding schedules and poor internet connections, only three interviews were conducted. Nevertheless, in light of the limited number of responses, additional interviewees were identified through a series of conversations and follow-up correspondence with informants. Recommendation of the interviewee and communication, led to 24 experts being reached out via email. Out of 24 contacted individuals, 15 interviews were conducted to ensure as diver sample as possible (Table 1). Likewise, sample size and data collection were adapted based on ongoing assessment of knowledge gaps through preliminary analysis of the interviews. A brief note about the research, interview purpose, timing, etc. was shared beforehand with the potential interviewees. Moreover, questions in the interview guide was also adapted as per the experience and involvement of participants at different level (Table 1). Interview invitation and interview guide is attached in the Appendices.

Category	Unit	Expected outcome	
		Insights on adaptation planning, financing, and other structural mechanisms	
Managerial level			
staff (donor		Perspective on what adaptation, maladaptation means in	
representative)	Country or regional office	practice	
	Lead INGO donor, country	Perspective on challenges and opportunities on	
	office staff, and local	implementation of adaptation project	
Operational level	implementing partner		
staff	organization	Insight on outcome of the projects	
		Broader perspective and conceptualisation on adaptation	
		pathways	
Academics and			
independent	Universities and think tank,	Inputs and suggestions for projects with low risk of	
practitioners	government unit	maladaptive outcome	

Tabla	1	Interview	gampling	ovitovia
rable	1-	Interview	sampting	crueria

Overall, the majority of the interviewees (ten individuals) represented the operational level, including community-based NGOs and INGOs. It should be noted that although all of them were not directly involved with the case studies, their insights were valuable. Four interviewees represented academia and worked as independent researchers/consultants, while one interviewee was from a multilateral aid organization. All interviews were conducted online using platforms such as Zoom and Microsoft Teams, following the interview guide. However, there was also room to explore emerging topics based on the interviewee's experience. The interviews were recorded directly and stored using pseudonyms to ensure confidentiality.

2.3 Data Analysis

All interviews were transcribed and audited to initiate a detailed analysis of primary data in NVivo, a qualitative data analysis tool with coding query mechanism (Bazeley & Jackson, 2007). The coding was performed in two phases. First, the transcribed interview script was scrutinized line-by-line to generate the initial open codes. This was to open up the inquiry and

make a better sense of what the interviewee was trying to say. This according to (Strauss, 1987, p. 31) is to answer what is actually happening in the data? And what is the main story in the dataset and why? No pre-existing framework/categories were applied during the initial coding but were left open to all possible emergent themes (Charmaz, 2006). This provided 83 provisional concepts and emergent ideas. To ensure that the analysis was not restricted to these initial codes, notes and memos were maintained alongside the areas of modification and elaboration as the analysis proceeded (Strauss, 1987). In the second round, transcripts were analysed in depth to determine the relationship between different ideas and potential categories. Once there seemed to be a saturation in the data (Grbich, 2013), codes were scrutinized to identify relations, and form a cluster of codes. Provisional themes identified and deemed irrelevant for interpretation were removed from further process. Lastly, analytical memos and inspiration from the literature facilitated the systematic coding of preliminary codes and eventually led to categorized codes and sub-codes. Findings from the interviews were also supplemented by analysis of project documents, including the project proposals, evaluation reports, etc. Lastly, emergent themes were validated by the literature, interpreted, and discussed in relation to the research purpose.

3. Conceptual background

This introduces concepts informing the research, supporting the analytical framework, and providing a foundation for the discussion of the findings. This chapter aims to provide and understand the central terminology used throughout the thesis.

3.1 Vulnerability

Climate interventions have long been guided by the concept of vulnerability, although it is known to be notoriously unclear and difficult to define (Adger, 2006). Vulnerability is a central concept in climate change research with wide array of definitions and interpretations (Füssel, 2007). In the past, vulnerability was conceptualized as a function of exposure and sensitivity to certain risks (Blaikie et al., 1994). This is more common in the actor-cantered approach, which largely relies on temporary events or risks that generate action in the form of preparedness or adaptation. However, the advancement in system-oriented approaches has led to a resilience framework that recognizes the dynamic complexity and nonlinear interaction of different components across scales (Câmpeanu & Fazey, 2014). More recently, efforts have been made to combine these two approaches, thereby rethinking vulnerabilities as structural properties deeply embedded within the system at multiple scales. Empirical evidence raises concerns about the systemic (re)production of differential vulnerability to climate change (Thomas & Warner, 2019). The most common framing of vulnerability in climate change literature is outcome vulnerability and contextual vulnerability (O'Brien et al., 2007). These are not just the different interpretation of word 'vulnerability' but the fundamentally different framing of the climate change problem. While outcome vulnerability is scientific framing, contextual vulnerability is human security framing (ibid). Outcome vulnerability is often perceived as a linear result of certain biophysical changes due to the impacts of climate change,

whereas contextual vulnerability is perceived to occur within contextual factors such as social, economic, and political structures (O'Brien et al., 2007). Therefore, vulnerability is not merely influenced by changing biophysical conditions, but through these dynamic interactions, resulting in compounded vulnerability (Simpson et al., 2023).

This research refers to UNDRR terminology "The conditions determined by physical, social, economic and environmental factors or processes which increase the susceptibility of an individual, a community, assets or systems to the impacts of hazards" (UNDRR, 2022). Vulnerability differs across scales, community by community, and individuals and households, despite exposure being the same. Vulnerability could be produced and reproduced over time between social groups within their lived environment and this could be referred to as 'vulnerabilization' (Chmutina et al., 2022). It can be enacted through climate change adaptation measures at multiple scales, such as structural conditions (e.g., poverty, inequality), institutional structures/policies, cultural norms/values. and individual characteristics/behaviours. This typology helps to understand how different types of vulnerabilities are produced and reproduced in order to better adapt to environmental changes like climate change or disasters. For the purpose of this thesis, contextual vulnerability forms an important base, as adaptation projects often need to be questioned as to why certain groups of people and communities are more or less vulnerable and affected by climate threats (Taylor et al., 2022). Hence, this shifts the focus beyond climate change alone to the underlying circumstances and the multiple processes of change. The interpretation of vulnerability directly influences the types of funded projects and how they are implemented. Adaptation projects and policies addressing 'outcome vulnerability' often lead to technocratic solutions that naturally ignore the underlying causes of vulnerability (Nagoda, 2015; Nightingale et al., 2020). In the worst case, they might entrench the existing pattern of vulnerability and even limit their adaptive capacity for future risks.

3.2 Climate change risk

Following this line of argument, this paper applies the contextual vulnerability approach and consider climate change as one of the causes of vulnerability among others. Thus, climate change risk is a result of the inherent dynamic and complex interaction of social, environmental, political, and economic factors. Environmental change is one of the many stressors that feeds into the dynamic interaction of the human environment system and reconfigures or exaggerates the existing patterns of vulnerability (Adam et al., 2018). So, this research focus on contextual framing of climate risk, meaning acknowledging the role of social and cultural processes in underlying framing of adaptation responses (Eriksen et al., 2015). The adaptation literature, to a great extent, recognizes the cascading effects of compound climate events, where one determinant of risk affects another, which in turn affects another in a linear manner (Simpson et al., 2023). This is referred to as the domino effect (ibid). Similar to the hegemonic framing of vulnerability, climate change risk discourse seems to be predominantly influenced by the biophysical emphasis of climate science (Ojha et al., 2016). Global narratives

on the technological framing of policy issues are accepted by countries in the Global South. Often, national adaptation plan addresses the technical experts, and international donors perspectives, and hardly touches upon real experience of vulnerable communities (Ensor et al., 2019; Ojha et al., 2016). Climate policies in the developing world have always been shaped by development agencies, donors, and service providers. One of the key reason behind this is financing mechanisms and knowledge production and legitimization around western scientific world views (Tabassum, 2022). How climate risk is framed in the wider discussion has its clear implications in the type of adaptation projects and how they are implemented.

IPCC (2022, p. 4) defines climate change risk as "the potential for adverse consequences for human or ecological systems, recognizing the diversity of values and objectives associated with such systems." Hazards, exposure, and vulnerability are key determinants of climate risk; however, they are subject to uncertainty in terms of magnitude and likelihood of occurrence (Blaikie et al., 1994). This thesis views climate change risk from holistic perspective to understand how the structural causes of vulnerability interact with ongoing and contextual realities of change.

3.3 Adaptation

Subjective notions of vulnerability are inherently drawn into adaptation, which is clearly observed in widely differing approaches (Klepp & Chavez-Rodriguez, 2018). Adaptation has a long and multidisciplinary history of study; hence, its meaning differs among various fields and practices (Brown, 2011; Eriksen & Brown, 2011). Given the highly contextual nature of adaptation, a widely accepted definition of effective means of adaptation is challenging. The concept of autonomous vs. planned adaptation is still reflected in the essence of incremental vs. transformative adaptation (Brown et al., 2017; Loginova & Batterbury, 2019). Incremental adaptation is usually perceived as an approach with sets of actions in response to environmental stress aimed at reducing exposure and sensitivity, or increasing coping capacity (Park et al., 2012) (Figure 2). Adaptations at this scale are reactive and short-term nature focused. In contrast, actions that are adopted on a larger scale, unique to a certain system, and those that transform places and shift locations are usually considered transformative adaptation (Kates et al., 2012). Such actions include fundamental changes in existing institutional arrangements, priorities, and norms (Park et al., 2012), expanding beyond the proximate causes of risk to its root causes, and ensuring sustainability in natural and human systems (Magnan et al., 2020). Incremental adaptation by nature addresses exposure and vulnerability, which eventually has short-term benefits for risk reduction but might turn out to be detrimental in the long run. While adaptation is usually perceived as a long-term perspective, it refers in practice to short-term coping measures, dealing with the immediate disturbances caused by external stress, such as climate change (Glover & Granberg, 2020). Along with the proliferation of adaptation literature, two major conceptualizations could be observed. One is adaptation as a response to externally imposed environmental change, and the other is adaptation as a part of sociopolitical processes that shape climate change and inequality (Nightingale et al., 2022).

"Adaptation involves changes in social-ecological systems in response to the actual and expected impacts of climate change in the context of interacting non-climatic changes. Adaptation strategies and actions can range from short-term coping to longer-term, deeper transformations, aim to meet more than climate change goals alone, and may or may not succeed in moderating harm or exploiting beneficial opportunities." (Moser & Ekstrom, 2010, p. 2026). This is one of the most comprehensive definitions and implies to wider literatures calling for considering adaptation as part of pathways of change and response, and not as outcome (Wise et al., 2014). This considers dynamic interactions among multiple stakeholders and their values, goals, and knowledge to avoid maladaptation and facilitate transformative strategies (Werners et al., 2021). Adaptation is beyond the mere response to climate stimuli; it is part of an existing human-environment system and vulnerability context determined by different factors (Adam et al., 2018).

This thesis is built upon the concept of 'situated adaptation' which means situating *climate* change within the complex, multidimensional, and transforming realities of marginalized populations (Ensor et al., 2019; Quealy & Yates, 2021, p. 2; Sen, 2023). This approach complements the concept of adaptation pathways (Wise et al., 2014). However, it also demands the need to recognize the local contexts and conditions in which adaptation takes place (Taylor et al., 2022). Adaptation is not a one-size-fits-all solution, and hence should be tailored to specific needs and priorities of the marginalized communities. Thus, rethinking adaptation through this approach might diversify actions by situating adaptation within the everyday realities of the people and communities, and shift away from the mainstream adaptation approach which is limited to tecno-managerial fixes (Juhola et al., 2022; Nightingale et al., 2020). This helps to step back and analyse how people are situated within the socio-natural context and how it co-produces differential vulnerability and benefits (Quealy & Yates, 2021). Likewise, this leads to the notion of 'vulnerabilization' (Chmutina et al., 2022) by revealing underlying socio-political dynamics and how people's lives are structured around this (Quealy & Yates, 2021). While there are wider calls for paradigm shift in adaptation research and action from moving away from technocentric mindset driven by western scientific knowledge and depoliticized solutions, 'situated' understanding of adaptation questions: "adaptation for whom"? (Adger et al., 2005; Atteridge & Remling, 2018; Ensor et al., 2019; Forsyth & McDermott, 2022). Adaptation is therefore part of wider processes of change and response situated within everyday experiences of vulnerability and constituted through humannonhuman relations. Comprehensive understanding of pathways of change for different groups and how they interact with each other is essential for future adaptation that is able to address both environmental and social justice related outcomes.

3.4 Maladaptation

Like adaptation, evolution of 'maladaptation' could be traced back to evolutionary biology, and then applied to different scholars including cultural, human, environment interface (Magnan et al., 2016). It is often used to describe unfavourable outcomes of adaptation efforts. There is no fine distinction on whether outcome could be labelled as maladaptive only in case of planned action or is it applicable to spontaneous response also (Jones et al., 2015; Mersha & van Laerhoven, 2018). Early maladaptation scholarship was quite straightforward, and the concept referred to adaptation efforts that increased vulnerability to climate change. Along with increasing interest and research, this concept has evolved elusively in its infancy. Factors complicating maladaptation, such as multiple drivers and geographical and temporal scales, are currently being discussed (Magnan et al., 2016; Strøbech & Bordon Rosa, 2020). Notwithstanding, there is still a significant difference between definitions and different opinions on how far and wide the adaptation effects should be analyzed in the future (Juhola et al., 2016).

The understanding of maladaptation through this thesis is consistent with the early conceptualization of IPCC (2001) as "any changes in natural or human systems that inadvertently increase vulnerability to climatic stimuli; an adaptation that does not succeed in reducing vulnerability but increases it instead" (p. 389). Although this definition does not refer to temporal and spatial scale issues, it has attracted the attention of academics and practitioners. Recent IPPC report with quite a focus on 'addressing maladaptation' shows that the concept is now in the mainstream adaptation literature, yet, little or no consensus on what it means in practice (IPCC, 2022). Simply, maladaptation refers to an action or decision taken by people, organizations, and governments which may have unintended negative consequences for their ability to adapt successfully over time. Maladaptation can occur when decisions are made without considering long-term impacts on ecosystems and communities; it often results from short-sightedness or lack of knowledge about potential risks associated with certain actions. In some cases, maladaptation arises due to external factors, such as economic pressure or political interests that override environmental considerations. To better illustrate what maladaptation looks like, one of the most commonly reported form is infrastructural maladaptation(Schipper, 2020). For example, sea walls built to protect people from rising sea level have made people more exposed by preventing stormwater drainage, and also shifting vulnerability to elsewhere along the coast (Piggott-McKellar et al., 2020). Other examples include large scale irrigation projects where many small holder farmers could not afford and sell or rent their croplands, agriculture intensification strategies leading to water pollution, increased greenhouse gas emissions, soil degradation, livelihood diversification strategies such as charcoal production resulting in increased deforestation, etc. (IPCC, 2022). The discourse on maladaptation goes beyond the current conception of adaptation which predominantly understands vulnerability to be driven by climate change rather than socio-economic stressors (Inderberg et al., 2014). So, this thesis considers 'maladaptation' as a tool to initiate critical reflection and discussion on externalities of adaptation decisions (Adger et al., 2005), and to holistically understand "what is being adapted to, Who implements, and How adaptation occurs?" (Asare-Nuamah et al., 2021, p. 2).

3.5 I/NGOs role in climate adaptation

Effective governance to reduce vulnerability to climate change now engages a range of actors including governments, civil society, and the private sector. Bilateral funds and large-scale multilateral funding schemes funded by the developed countries are the major gateway for adaptation actions and are often implemented by local governments and NGOs. Historically, international actors have instrumental influence over developing countries through foreign aid (Dolšak & Prakash, 2018). Despite the decades of effort and investment, government and aid agencies still fail to effectively reduce the vulnerability of the marginalized (Tiwari, 2015). However, during this discourse, our understanding of vulnerability changed significantly, at least in the literature. But this has not been translated into practice among organizations. Project interventions are predominantly shaped by technocratic and depoliticized perspectives, which can also be traced back to the natural and climate science disciplines of adaptation (Brink et al., 2023). In fact, international donors want to stay away from the deep discussions and do not explicitly tackle the context specific political and power dynamics. In most cases, this is undesirable for them. Instead, they prefer to opt for and promote technocratic, apolitical, and outcome-oriented approaches, which tend to be legitimized through the process of so-called community participation (Buggy & McNamara, 2016; Nagoda & Nightingale, 2017). This is a safer way for the organizations to pursue quick and measurable results (ibid). This might not lead to adverse impacts in all cases but undermines the long-term ability of the respective government to address the real and vulnerability needs of its citizens. This legitimizes expertdriven climate change knowledge, and projects led by I/NGOs directly or indirectly prioritize this mindset at the expense of contextual needs, realities, and local knowledge (Tabassum, 2022; Ziervogel et al., 2022). In other words, organizations driven by donor mindset, are interested in producing quick and visible results within a limited time frame, without much capacity or instrument to design radical shifts in social and power relations within CCA at the centre of the projects (Nagoda & Nightingale, 2017). Along with the growing attention to locally led adaptation and climate justice agendas, aid organisations have a greater role to play (Westoby et al., 2020). There is growing investment to building voice and agency and supporting active engagement of community. This thesis acknowledges their role in bridging the gap between international climate funds, national executing entities, local partners, and community people.

4. Literature review

This chapter aims to provide brief overview on how key concepts of the research (adaptation and maladaptation typologies) are framed in the wider literatures. This helps to develop a general understanding of the concepts and presents this thesis' interpretation of these concepts, as these are often ambitiously defined and used. Likewise, this chapter also identifies and summaries the key frameworks and guidelines on conceptualising maladaptation and draws key insights and provides the building blocks to understand the research standpoint.

4.1 Adaptation framing

Recent critiques of adaptation policies and their outcomes have led to a shift in adaptation framing. This is evident in the literature that, when adaptation is framed as an environmental problem response, it invites technocratic and managerial solutions (Eriksen et al., 2015; Glover & Granberg, 2020). Such measures are often highly risky in terms of maladaptive outcomes. So, prevailing technocratic vision is subjected to critiques among adaptation scholars, which has revealed the underlying socio-political and economic implications (Inderberg et al., 2014). Adaptation actions without considering the social power relations and the socio-structural constraints widens the vulnerability gap rather than reducing it (Browne & Razafiarimanana, 2022; Fünfgeld & Schmid, 2020). Adaptation does not operate in silos, but rather occur in the context of socio-economic, cultural, and other changes brought about by globalization. These are non-climatic factors leading to adaptation, and it is difficult to clearly distinguish between climate change adaptation decisions and actions in response to non-climatic factors (Adger et al., 2005). Therefore, attributing action to climate change is not a simple process. One of the most common framing or principles of effective adaptation is 'transformation framing,' meaning "adaptation process should fundamentally change human thinking and practices in the face of climate change and overtly challenges the power structures that generate vulnerability" (Singh et al., 2022, p. 656). Transformation framing is generally based on the assumption that fundamental changes are feasible and essential (Schipper, 2020). Therefore, the adaptation process should truly challenge existing power structures that generate vulnerability in the first place (Kates et al., 2012) (Figure 2). Meanwhile, transformation in relation to climate change has been a hot topic for discussion based on the understanding that climate change poses risks that are beyond the ability of human environment system through business as usual approach, in other words, incremental adaptation (Uitto & Batra, 2022). Just adaptation is clearly insufficient for the society to manage complex climate risks.



- MALADAPTIVE OUTCOME -------

Figure 2: Illustration inspired and adapted from literatures (Bassett & Fogelman, 2013; Moser & Ekstrom, 2010; Pelling et al., 2015)

Mainstream adaptation practices, which are predominantly coping and incremental, focus on the biophysical impacts of climate change and measures to cope with the adverse impacts. The conceptualization of such adaptation approaches (figure 2) is not always distinct and requires an understanding of different priorities. In addition, any adaptation action that is labelled coping or incremental might lead to transformational action (Wilson et al., 2020). It is evident that adaptation is quite a slippery concept, as it is broadly interpreted, which means different things to different people. However, impact-led conceptualizations are significant in practice, which view adaptation as an adjustment. And non-governmental organizations are not an exception to this. The adaptation projects implemented by I/NGOs mainly focus on responding to the immediate impacts of climate variability and helping communities to cope with extreme events. Similarly, in the academic literatures, Bassett & Fogelman (2013) reported that 70% of the articles perceive climate impacts as the major source of vulnerability whereas only 3% consider deeper roots for vulnerability and call for transformative adaptation. In recent years, there has been wider acceptance that risk should be located within societal and biophysical hazard interactions, which successively promote transformational adaptation (Quealy & Yates, 2021). Most evidently, heavily technocratic inclination of adaptation literature and practice is being critiqued through the political nature of adaptation (Dolšak & Prakash, 2018; Glover & Granberg, 2020; Tabassum, 2022).

4.2 Maladaptation and its typologies

In climate and development literature, maladaptation is a relatively recent term that has gained widespread attention among academics and practitioners. The earliest introduction of maladaptation in the climate adaptation sphere mentioned that "Maladaptation can result in negative effects that are as serious as the climate-induced effects being avoided (Scheraga & Grambsch, 1998, p. 87). Unsuccessful adaptation does not necessarily have to be maladaptive unless it results in adverse effects (Barnett & O'Neill, 2010). The key link is vulnerability, which refers to an increase in the vulnerability of a system or people. As argued by (Lama et al., 2017), labelling certain actions as adaptive or maladaptive is highly normative, value laden, time and context dependent. Boutroue et al., (2022) further stressed that "(mal)adaptation is as much a political process as a scientific endeavour" (p.893). This is because it is contested within socially situated actors and their values (Fünfgeld & Schmid, 2020), which cannot be constructed using scientific knowledge alone. There is consensus in the literature that proposing an adaptive action as maladaptive or transformative is not straightforward and demands holistic and long-term pathway understanding (Brink et al., 2023; Work et al., 2019). Any intended action and its success or failure might change over time as the values to prevent something change with the continuous interaction of the human environment system and its components (Ayanlade et al., 2023).

Although there is consensus that adaptations to climate variability might escalate ingrained vulnerability or even introduce new sources of vulnerability (Torabi et al., 2018), conceptual ambiguity remains in defining such an outcome. In addition, the majority of the conceptualization (Table 2) provides room for a certain degree of subjectivity and vague interpretation, leaving the questions of Who? How? Why? and other questions regarding adaptation actions and outcomes.

Literature source	Maladaptation defined
(IPCC, 2022, p. 6)	"any actions that may lead to increased risk of
	adverse climate related outcomes, including via
	increased greenhouse gas emissions, increased or
	shifted vulnerability to climate change, more
	inequitable outcomes, or diminished welfare, now or
	in the future"
(Glover & Granberg, 2021, p. 69)	<i>"adaptation actions, plans, policies and processes</i>
	intended to avoid or mitigate vulnerability to climate
	change that has, or is predicted to have, negative
	impacts on individuals, communities and/or systems"
(Ma et al., 2021, p. 4)	<i>"actions that may alleviate short-term and individual</i>
	impacts, but would deter the building of long-term
	sustainability and go against public adaptation."
(Juhola et al., 2016, p. 139)	"result of an intentional adaptation policy or measure
	directly increasing vulnerability for the targeted
	and/or external actor(s), and/or eroding preconditions
	for sustainable development by indirectly increasing
	society's vulnerability"
(Barnett et al., 2013, p. 226;	"action taken ostensibly to avoid or reduce
Barnett & O'Neill, 2010, p. 211)	vulnerability to climate change that impacts adversely
	on, or increases the vulnerability of other systems,
	sectors or social groups"
(Magnan, 2014, p. 3)	"process that results in increased vulnerability to
	climate variability and change, directly or indirectly,
	and/or significantly undermines capacities or
	opportunities for present and future adaptation"
(OECD, 2009, p. 53)	"business-as-usual development which, by overlooking
	climate change impacts, inadvertently increases
	exposure and/or vulnerability to climate change"
(Global Center on Adaptation,	"when an intervention to adapt to climate change
2022, p. 239)	makes people more vulnerable"
(Mycoo, 2014, p. 137)	"intervention in one location or sector may increase
	the vulnerability of another location or sector, or

Table 2: Key definitions of maladaptation in the literature, inspired and adapted from (Chi et al., 2021, p. 11)

	<i>increase the vulnerability of the target group to future climate change</i> "
(UNFCCC, 2007, p. 32)	<i>"adaptation measures that do not succeed in reducing vulnerability but increase it instead"</i>
(Yaro et al., 2015, p. 3)	"negative changes and practices resorted to by households and individuals in reaction to climate stressors that are inimical to their welfare or that of the community as a whole"
(Fenton et al., 2017, p. 193)	<i>"maladaptation occurs if exposure of sensitivity to natural hazards and stress is inadvertently increased or adaptive capacity reduced"</i>

Although it is an ambiguously defined concept, in the broad notion, maladaptation is largely perceived as the counter-narrative of what adaptation aims to accomplish. Based on (Schipper, 2020) conceptualisation that maladaptation must be understood as a phenomenon along the continuum from success to failure, (Glover & Granberg, 2021) emphasize that maladaptation is a complex, contested, and difficult concept. What makes this complex phenomenon is the wide differences in factors, such as framing, assessment criteria, nuances, and values (Glover & Granberg, 2021; Magnan, 2014). The concept of maladaptation informs and prompts essential critiques of adaptation. "*What are the results of adaptation, including its unintended consequences? How do we evaluate these outcomes*?" (Glover & Granberg, 2021, p. 70).

4.3 Existing frameworks and tools on understanding maladaptation

One of the earliest frameworks for conceptualizing maladaptation was by (Barnett & O'Neill, 2010) which identified five distinct pathways; increasing in greenhouse gas emission, increasing vulnerability of groups most at risk, high socio-economic or environmental cost, reducing incentive to adapt, and path dependency. Advancing on this, (Magnan, 2014) adopted Pathways Framework with the principle of avoiding maladaptation. This includes ensuring that adaptation promotes socio-economically equitable interventions, cost-effective initiatives, flexibility, develops incentives to adapt, and does not lead to increased greenhouse gas emissions (Barnett & O'Neill, 2010; Magnan, 2014). This understanding has progressed along with increasing evidence of adaptation actions leading to unintended outcomes (Guodaar et al., 2020; Piggott-McKellar et al., 2020). Advancing on the maladaptation literature (Jones et al., 2015) tried to define three distinct types of adaptation outcomes; successful adaptation, failed adaptation, and maladaptation. Like many conceptualizations of maladaptation, these outcomes are highly value-laden, normative in nature, and vary according to individual interpretations (Boutroue et al., 2022). Nonetheless, this provides a basis for researchers and practitioners to relate to potential pathways. They define successful adaptation strategy as "one with either significant or limited positive effects across collective climate risk, distribution of risk over time, and no negative impacts on well-being" (Jones et al., 2015, p. 10) (Figure 3). Similarly, an adaptation strategy could be considered to fail when "it has negligible impact (neither positive nor negative) on reducing climate risk now and/or in the future" (p.10). This

leads to defining maladaptation as a strategy that has a negative impact on the climate risk of well-being. Contributing to the burgeoning literature (Juhola et al., 2016) identified three typologies of maladaptation; rebounding vulnerability, shifting vulnerability, eroding sustainable development. In contrast to the previous framework, they argued that 'spontaneous adaptation action' should not be included in the analysis of maladaptation, as adaptation is often a by-product in such cases. This addressed the existing dilemma of drawing system boundaries and operationalizing the concept of maladaptation. This would strengthen the conceptualization, but this is still an ongoing discussion on how autonomous adaptation, for example, spontaneous action taken by a household, could lead to adverse impacts at the community level (Mersha & van Laerhoven, 2018; Mycoo, 2014; Rahman & Hickey, 2019).



Figure 3: Illustration inspired and adapted from (Jones et al., 2015; Juhola et al., 2016; Mills-Novoa, 2023)

Although these classifications (figure 3) may have some practical ambiguity, they offer a comprehensive understanding and foundation for examining the interconnectedness between them. Simpson et al. (2023) analysed 45 different types of responses across 39 countries and found that 41% showed maladaptive characteristics. There is limited evidence showing that adaptation response reducing risk, instead, inappropriate responses leading to increased vulnerability (Atteridge & Remling, 2018). Despite the growing concern in academia and international forums, this has not been translated in practice, especially among the projects designed and implemented by INGOs. This is partly due to the lack of frameworks and guiding tools for practitioners to help communities respond to the immediate impacts of climate change

variability while ensuring long-term sustainability and benefits of their actions. Existing frameworks do not really help development practitioners to facilitate the design of more effective and sustainable projects with a low risk of maladaptive outcomes (Hallegatte et al., 2020; Stockholm Environment Institute, 2021). One of the common and central aspects of the ongoing discussion is 'vulnerability' (Figure 3). Three patterns of influence on vulnerability determine pathways from adaptive to maladaptive outcomes; reinforcing existing vulnerability, redistributing vulnerability, and introducing new sources of vulnerability (Eriksen et al., 2021). Reinforcement of existing vulnerability occurs when adaptation goals and priorities are set by relatively privileged groups, as this results in an inequitable distribution of benefits and resources in favour of limited elite groups (Eriksen et al., 2021). This phenomenon is referred to as 'elite capture', and has been a long-standing problem in development (ibid). Such a situation ultimately reinforces existing power relations and marginalisation, which are the drivers of differential vulnerability and inequality in the first place (Thomas et al., 2019). Owusu-Daaku (2018) introduced the concept of (mal)adaptation opportunism to refer to situations in which adaptation projects are driven by certain economic interests rather than intended objectives. Likewise redistributing vulnerability over a broader spatial area or among other groups is often reported in infrastructural and technical interventions, for instance, hydroelectric dam, flood embankments, coastal infrastructures, and agro-technologies (Eriksen et al., 2021). Lastly, introduction of new risks and sources of vulnerability results from adaptation measures that emphasize short-term concerns and inadvertently creating long-term risk. This involves the trade-offs between short-term coping and reducing future risk, thereby resulting in unsustainable initiatives that create negative path dependencies in the long term (Eriksen et al., 2021).

5. Case study descriptions

These case studies are projects drawn from NGOs and their community-based partners operating in Zambia, India, and Ethiopia. The projects were typically two- or three-year-long projects implemented between 2016 and 2023. It should be noted that they are situated within a broader programme of community development, responding to local area development plans (those of government, and as identified through participatory planning processes involving the community). These projects form the basis for questioning the international NGO donor, the country level NGOs, local partner organizations and other stakeholders involved in their implementation on the concept of maladaptation.

5.1 Right-based project with an agricultural component, Ethiopia

This was a two-year project that focused on multi-level coordination involving the contribution of children, mitigating, and preventing the immediate and downstream effects of climate change and environmental degradation in five kebeles in Ethiopia. The overall project had five objectives, with two focused on children specifically (school safety and child protection) and three more aligned with the subject of this thesis: Increased food security in vulnerable households through improved and diversified income opportunities; Strengthened household resilience to prepare for and cope with disasters by improved and diversified agricultural production; and environmental stewardship and strengthening service providers linking the climate change and child protection nexus.

5.2 Gender empowerment project supporting climate smart agriculture and economic development in India

This project included supporting women in remote parts of rural India to establish a climate informed, economically viable, democratic, and self-governing business enterprise, with the support of Civil Society Organisations. The project supported the creation of women farmer groups and farmer producer companies, leading to improved productivity, better market selling prices and access to information and communication. Climate smart agriculture was promoted and linkages to markets were formed, as well as connecting the women farmers with agriculture investments and programs.

5.3 An integrated agricultural project with a larger water component, Zambia

The goal of this project was to promote sustainable economic development and food security at household level in an area of rural Zambia that was plagued by drought, floods and wild animal incursion. Six main components were identified to achieve the outcomes of the project; crop diversification and agroforestry, dam rehabilitation and irrigation (rain water run-off catchments, hydro-powered pumping from the river and boreholes), fish and livestock farming, solar powered electric fencing to deter animal invasion, climate change and disaster risk reduction preparedness training to farmers, and agricultural marketing training. Three main outcomes were:

- sufficient water is available to irrigate crops and trees
- people raise livestock and fish, and grow diversified crops, fruit, and agroforestry trees
- crops and trees are protected from animals

6. Results and Discussion

Data analysis process as described in section 2.3 yielded 8 overall themes comprising 4-5 sub themes under each. This chapter will simultaneously present and discuss those findings in relation to overall purpose and research questions. Considering the limited number of interviews, this chapter also refers to emergent aspects through triangulation. This section also interprets the research findings in relation to conceptual background and review of established literature as discussed in section 4. The figure below provides quick summary for what it means to reduce the risk of maladaptation in future adaptation projects among development organizations and followed by in depth analysis of key factors identified by this research.



Figure 4: Insights for future planning and implementation of adaptation projects based on interviews and guiding principles of locally led adaptation

6.1 Understanding context

Contextual understanding during the design and implementation phases of the project was found to be the key factor influencing the success or failure of any project. The majority of the interviewees highlighted the potential impacts that might arise when a project fails to acknowledge such context specific factors, interviewee 8 stating that "projects operating at local scale should understand the contextual dynamics in enough detail to be able to understand how that is adversely affecting the community (Interviewee 8, donor representative)." Such contextual factors include behavioural and cultural aspects, differential vulnerability, ground reality, poverty and competing needs, and power dynamics. Empirical evidence in the growing body of adaptation literature demonstrates how internationally funded interventions often overlook such factors and socio-political relations and processes (Eriksen et al., 2021; Gaworek-Michalczenia et al., 2022; Nunn et al., 2020). This is not only with contextual vulnerability, but also with differential needs and capacity, as interviewee consistently highlighted the importance of "participatory need assessment." An inadequate understanding of these contextual aspects has direct implications for addressing the drivers of vulnerability, which is a key characteristic of effective adaptation (Singh et al., 2022). To ensure that adaptation interventions are effective and do not increase the vulnerability of those they seek to help, it is important to consider the social divisions that exist in society and how they shape climate change adaptation (Pritchard & Thielemans, 2014). This includes understanding how gender, race, age, (dis)ability or class can determine who is vulnerable to climate change and who has a greater ability to adapt. Some informants also reported that cultural beliefs, norms, knowledge and attitudes play a vital role in successful implementation of adaptation options but unfortunately most organizations do not pay much attention to this. One interviewee shared their experience, stating that "it was not easy to convince some

communities who value cattle as a measure of wealth and respect, and switch to keeping small animals that are initially looked upon as those for poor families (Interviewee 6, livelihood and adaptation expert)." This implies livelihood diversification strategies, which are often a significant component of adaptation projects (Antwi-Agyei et al., 2018). So, projects must be sensitive to more than just presenting practical solutions. Interviewee working in similar contexts also indicated that investment in adaptation projects focused on the livelihood and agriculture sector has yielded benefits, but these agricultural techniques and practices have not always continued and scaled up to other communities. This result supports previous research that has referred to such vulnerability linked to the productivity-vulnerability paradox (Quealy & Yates, 2021, p. 4). This means that many smallholder farmers supported through internationally funded projects are made more vulnerable due to the modernized agricultural practices and techniques used to increase productivity. Although increased agricultural productivity has short-term benefits, it may make smallholder farmers more vulnerable by trapping them in an unsustainable debt cycle, eroding collective norms, and even ecological failure in some cases (Quealy & Yates, 2021). The case study projects (Section 5) also shows similar patterns of livelihood support measures, and it is possible that the paradox might deepen the uneven distribution of risks and benefits. For example, projects (5.2, 5.3) promote shifting from traditional to high-value commercialized crops as a means of livelihood diversification but do not reflect on the volatile price, market dynamics and access, increased investment, competition, etc. Therefore, farmers who benefit from such interventions are the ones who are already well resourced. The literature suggests that such measures do not necessarily lead to resilient livelihoods in all contexts and, hence, need further research (Acharya et al., 2021). This was also evident in the interviews, where interviewees consistently mentioned the unsustainable promotion of cash crops, resulting in limited or no impact on the economic benefit of the farmers in the long-term.

Additionally, such strategies are not always adopted by communities beyond the project target area. This is most likely because the project interventions are not tailored towards cultural elements and supporting behavioural learning. The case study projects also revealed similar findings and suggested that organizations should be careful in dealing with the behavioural and cultural aspects of the communities they work with. For instance, the women empowerment project in India had indicators such as 'transference of land ownership from male family member or husbands to female.' This has benefits and incentives in terms of different government schemes and subsidies specifically aimed at female farmers. However, specifying this as an outcome is not sufficient. It must be supported by actions that address the deeply embedded social and cultural beliefs attached to this, which might require generations to change. Projects must consider that change of this magnitude takes time and in the confines of a two- or three-year project, progress is likely to be small. Timely and patient financing is needed to achieve such results. Another interviewee emphasized that I/NGOs "very often use the same kind of approach that they view somewhere else, (Interviewee 1, adaptation researcher)" but the main priority should be contextual needs and priorities. In addition to the factors mentioned above, context-specific climate information is key. Projects should be well informed of the current and future climate variability, geographical nature of the environment,

adaptive capacities of local ecosystems, and so on. However, this varies from project to project, depending on nature.

Another key determining factor is the poverty and competing needs of communities. The analysis shows that these factors are often mentioned in the project documents but do not explicitly discuss their implications for project interventions and their outcomes. For instance, interviewee involved with project in Ethiopia stated that "although communities embrace the project interventions, we find that people still go to bush to cut trees for charcoal production (Interviewee 5, community-based NGO practitioner)" because they need food for their families, and they need to pay their children's school fees. This is where sometimes organizations come up with projects targeting immediate needs but may unintentionally prohibit the future adaptive capacity of individuals and communities, as the projects do not fully provide the solutions or may not have the funds to do so within the limited timeframe. Likewise, power and politics constitute another essential factor for understanding the context. The literature suggests that power relations are drivers of differential vulnerability patterns, as this influences unequal access to resources and decision making (Thomas et al., 2019). Thus, in the long-term, it enhances or hinders the adaptive capacity. If such structures and relations that led to the marginalization of individuals and communities are overlooked in ongoing CCA projects, it runs the risk of reproducing development, which contributes to vulnerability in the first place. It is important for practitioners and organizations to ensure that they are not lending themselves to inherent power dynamics, yet working closely with people trusted in the communities. This is challenging for the organizations. Advancing on how challenging it is to address these problems, one interviewee mentioned "it's precisely because of their lack of power that they are the most vulnerable and marginalized. And so, they're not going to show up to those project interventions or meetings. So how do you then try to engage and find out what exactly is necessary for them (Interviewee 1, independent researcher)?" This statement shows the complexity, but the key thing is to be aware of the underlying power distribution and its influence while making decisions on adaptation actions (Adger et al., 2005; Dolšak & Prakash, 2018).

Although certain funding guidelines incorporate power analysis and principles such as 'do no harm', a limited number of interviewees (researchers and donor representatives) mentioned its implications for projects. As suggested by one interviewee, "*such consultations and analysis might be one of the buzzwords; ticking the boxes, and number of workshops.*" However, community-based organizations play a vital role in ensuring that everyone is involved in collaborative and fair ways, as they negotiate power dynamics. Since the projects rely on existing structures and institutions for implementation, they simply follow established power relations that have historically marginalized certain people or groups. Therefore, projects should consider that interventions do not reinforce these inherent relations. This need should be made more visible in project documents to ensure that it remains acknowledged. This is evident in the literature which indicates that pre-existing power structures act on all climate change adaptation programs, excluding the marginalized people (Nagoda & Nightingale, 2017; Nightingale, 2017), eventually reinforcing existing inequality (Eriksen et al., 2021) and leading to maladaptive outcome. Projects ultimately help promote the interest of certain elites and

maintain their social status because of their pre-existing networks with politicians and easy access to local government and non-government authorities. Thus, most of the time, people benefiting from project interventions are the same people from within the limited networks (Nagoda & Nightingale, 2017). Some interviewees also accepted the fact that existing power relations prohibit the decision-making influence of marginalized people, but they feel limited within the trap of mandate, tools, or guidance to analyze and change the causes of exclusion in their work.

6.2 Equity and Justice

The majority of the interviewees indicated that justice and equity should be at the center of adaptation planning, project design, implementation, and evaluation (Figure 4). This means that considering how a certain person's adaptation may be accomplished at the cost of others increased vulnerability. As mentioned in the projects documents, the main goal of INGOs is to support the most deprived, excluded, and vulnerable. A common response was "we are concerned with marginalized and disadvantaged communities." Most interviewees embraced the 'justice and equity perspective', with one interviewee explicitly stating that "we can see the tilting of benefits towards people who are capacitated already, whereas people that do not, tend to be overlooked in multiple ways." This could also be related to the underlying issues of power and politics, as discussed in (section 6.1). When inequality and injustice patterns such as colonialism and racism are not explicitly recognized by a project, they limit the opportunity to resolve the root causes of vulnerability (Juhola et al., 2022). This renders entrenched structural disparities through which individuals and communities are marginalized (Fünfgeld & Schmid, 2020). The findings suggest that effective adaptation must start with an understanding of the distribution of risk, and adaptation projects cannot adopt a risk-blind approach. Therefore, every project stakeholder should continuously reflect on why certain individuals and groups of people face a greater likelihood of being exposed to and impacted by climate risk in a disproportionate manner (Thomas et al., 2019). Although the topic of sociopolitical analysis was not central to many of the interviewees, a few expressed the need for a strong political economy analysis while designing an adaptation project. It helps practitioners to be aware of conflicts and dynamics associated and ensure that certain groups are not negatively and disproportionately affected by the project intervention. Different forms of inequalities make it difficult for certain groups of people to navigate around social hierarchies and power dynamics and eventually have profound impacts on their exposure, sensitivity, and adaptive capacity. For instance, interventions such as in Zambia project, focusing on water resource management must consider distribution and access to water. Projects should explicitly identify the most in-need of the intervention and pre-sensitise the community as to why this group were chosen. This helps avoid conflict and ensure that projects interventions are not reinforcing vulnerabilities by reflecting on who benefits from the intervention. For example, people who have large low-lying land benefit more from water irrigation schemes, and people with high landholdings in accessible places benefit from the historical system of oppression.

One way of avoiding this is to ensure that there is meaningful engagement from the community. As some interviewees expressed their concerns "there is a difference between inclusion and *meaningful participation*". Research has shown that the participation of marginalized people in project interventions is often just for the sake of participation (Nagoda & Nightingale, 2017). This is to report on the number of participants and show on paper that the project has better community participation. However, the voices, concerns, and aspirations of the marginalized are not considered in planning, decision making, implementation, and management. Thus, practitioners should reflect on whether the participation of certain groups or communities is meaningful. The examination of projects documents indicated that all projects view certain groups of people, such as children, youth, and women, as homogenous entities and engage them in project interventions. However, they are not explicit about how they are connected to other actors and systems. Effective community engagement involves all stages of the process: identifying issues, designing responses, implementing actions, and evaluating results (Simon et al., 2020). This is important because it helps to ensure that the process is both scientifically sound and socially robust (Tye & Suarez, 2021). Involving community groups in the process also helps to promote accountability among experts and policymakers and helps overcome the deficiencies of top-down decision-making. One imperative yet neglected aspect in the research and practice dialogue of adaptation is the epistemic injustice (Tabassum, 2022) questioning, what the projects aim to achieve through so-called participation? (Forsyth & McDermott, 2022). It appears that the notion of the community being used hides social divisions. Viewing adaptation process through 'justice and equity' lens requires "deep co-production" around what is considered climate risk and who is impacted? (Forsyth & McDermott, 2022, p. 2). Moreover, this leads to transformational adaptation, which goes beyond implementing solutions for the predefined concepts of risk and vulnerable groups. Therefore, adaptation planning should be beyond the passive response to external threats and focus on changing social relations and historical injustices (Fünfgeld & Schmid, 2020). Who decides on the adaptation to take and who wins and who loses from it? should be consistently reflected (Nightingale et al., 2022) (Table 3).

6.3 Inadequate monitoring and evaluation

The review of projects documents suggests that the existing practice of monitoring and evaluation frameworks are not sufficiently equipped to assess the positive or negative effects of adaptation interventions on vulnerability. The current mechanism primarily focuses on project efficiency, which emphasizes how well the projects are implemented, rather than on their effectiveness and equity. M&E frameworks are highly focused on outputs, (for example, services delivered, number of beneficiaries reached, etc.) with little or no attention paid to long-term outcomes in the resilience or non-beneficiary population. MEL frameworks and instruments should be sufficiently flexible to allow programs to be constantly reviewed and adapted (Coger et al., 2021), especially for resilience building programs, which by nature are working with uncertainties and responding to emergent contexts and changes (Villanueva et al., 2018). They should not be static and should be tailored to project implementation as it progresses through its lifecycle. Technically, M&E frameworks do not always fit the climate

context, implying that long-term climate risk is not necessarily accounted for. Likewise, the evaluation of how and to what extent adaptation actions affect socio-political relations, resilience, development and adaptation gaps/needs are key limitations in existing practice (Bours et al., 2014). Simply put, M&E frameworks do not always interrogate negative or unwanted outcomes of adaptation projects and are typically formatted to report effective management of planned activities. The uniform set of indicators for all groups in the M&E framework contrasts with the differentiated priorities, risks, and impacts among different groups. Therefore, there is a dire need to shift towards realist evaluation, which acknowledges that contextual factors play a vital role in project outcomes and that no intervention works everywhere, or for everyone (Villanueva et al., 2018). This helps draw lessons on what works, where, how, why, and for whom. According to the community-based NGOs interviewees, projects are often evaluated by external consultants and perceived as a formality for accountability to donors. Such evaluation instruments often use measurable outputs so that funders can track the return on their investments. This is also essential to maintain the integrity of the climate finance mechanism (Green Climate Fund, 2021); however, it may lead to misleading conclusions, as project outputs do not necessarily result in impacts. Likewise, the output-oriented assessment of projects overemphasizes the manifesting impact simplistic framework and metrics such as the number of beneficiaries reached. Researchers argue that such success versus failure is overly simplistic in measuring adaptation outcomes (Canales et al., 2023; Mills-Novoa, 2023). This leads to projects with interventions that target a large number of beneficiaries yet have very little meaningful impact. It is essential to shift away from traditional tools and frameworks focused on performance assessment in order to generate evidence for learning (Taylor et al., 2022). Thus, such an 'evaluative monitoring' approach helps understand how and why change is happening (Villanueva et al., 2018). As reported by one interviewee, "evaluations that are done within the project timeframe, do not identify the real impact of these interventions (Interviewee 12, community-based NGO)." Current M&E tools are not equipped with time and resources in a way they would monitor the project as they unfold. Most importantly, the review of M&E frameworks suggest that, these rarely acknowledge that these systems are dynamic and so these tools are not adaptive to the dynamic context in which projects are operating in (GIZ, 2014; Nick Brooks, 2014).

Despite these challenges, practitioners at all levels agree that "there is a need for co-creation of M&E tools, with the users on board, the people who get to use the resources and benefits of the project." The community should decide what should be monitored, collected, and analyzed, which eventually leads to adaptive management processes and a better learning culture (further discussed in section 6.6) (Coger et al., 2022). Similarly, interviewee associated with case study projects expressed their concerns regarding the post-evaluation process, and the main problem is that it is not being built into the project design. While this does not necessarily lead to maladaptive outcomes, it is one of the key characteristics of an effective project. Therefore, long-term program thinking, monitoring and evaluation are necessary for effective adaptation projects. Thus, organizations should rethink rigid tools with a narrow focus on output and indicators, and maladaptation could be an entry point for further discussion.

Previous findings have shown that existing tools in M&E and their applications for effective planning, monitoring, and evaluation of climate change adaptation projects suffer from social desirability bias and acquiescence bias (Wojewska et al., 2021). During the surveys, KIIs, and FGDs, respondents to the M&E assignments are likely to overemphasize the positive views on the project, hoping to be in a favourable light. Similarly, they tend to agree with the questions rather than provide honest opinions. Resilience programming is all about working with uncertainties, embracing them, and being adaptive to the emergent dynamics as the program evolves through its lifecycle (Woodhill & Millican, 2023). Hence, rigid M&E frameworks must be tailored and responsive to the context in which the project is being implemented. Notwithstanding the aforementioned benefits of the MEL system, it is essential to recognize that it requires continuous reflection to serve the dual purpose of accountability and learning (Table 3). As a result, project teams and practitioners must be aware of and explicit about this from the outset.

6.4 Funding structure and mechanisms

The analysis shows that along with the wider call for locally led adaptation principles, structural issues with finance mechanisms are gaining attention. A significant majority of the interviewee highlighted the need for localized funding based on contextual needs and priorities. While this was mentioned as a criterion for a successful adaptation project, in reality, it is always difficult for INGOs and local partners to access finance through these bureaucratic regulations and requirements. According to one interviewee, "*it is too complex for community-based organizations to live up to these requirements and access to funds (Interviewee 9, community-based NGO)*." As a result of which, despite the huge investment in climate adaptation, it is rarely reaching the grassroot level in a way that allows the end users to make decision based on their need and understanding.

Although climate finance should prioritize the needs and priorities of those who are most affected by climate change (Holland et al., 2022), Browne & Razafiarimanana (2022) discovered that adaptation finance disproportionately favoured households that were already capacitated to adapt, rather than those in need of aid (6.2). Political connection with authorities holding power was found to be the main factor which unpacks the need for advanced approach to power and inequality in internationally financed adaptation (6.1). If projects continue to shift and redistribute vulnerability, it will eventually lead to greater risk and vulnerabilities for those who are already marginalized (ibid). In addition, neglecting the potential of maladaptive outcome, projects might make the situation worse. Thus, the inflexible and insufficient tools identified previously are in part a consequence of the funding structure and mechanisms in place. One interviewee expressed "it is quite frustrating because we have to do it again and again, as we are stuck in this system that requires us to do this in order to secure funding (Interviewee 15, INGO representative)." Advancing on this, another independent practitioner emphasized that "we should ask ourselves Who and What are we actually funding rather than this aid priority." Structural issues in the development sector, such as bureaucratic or systemic limitations on financing and reporting, often constrain projects to narrow conceptions and

short-term outcomes. One local partner representative stated that "when we put simple and realistic outcomes, they are not funded as this is very common for the donors". This reflects the previous discussion on overemphasizing positive outcomes, where practical and achievable goals may be undervalued. Another interviewee shared a similar perspective, mentioning that INGOs frequently overemphasize positive outcomes, rather than acknowledging errors and mistakes.

International adaptation finance often overlooks the dynamic nature of vulnerability and aggregates it at the country or community level. Multilateral and bilateral funders must acknowledge the entrenched inequalities and informal political dynamics that exist within the local context to avoid reinforcing or exacerbating existing inequalities through adaptation interventions. One interviewee highlighted the need for better coordination between different funds to achieve effective results and to understand the potential adverse outcomes before scaling up projects. However, there are fundamental issues within the development practice, and as one interviewee noted, "there is not much that an NGO or INGO can do" to address these issues. One of such conceptual issue in the structure is 'additionality' (European Parliament, 2012). Critical adaptation scholars argue that principles like additionality can be challenging, particularly when development and DRR actors have co-opted adaptation (Inderberg et al., 2014). Some interviewees noted that this has resulted in adaptation projects addressing only surface level impacts, rather than addressing the underlying drivers of vulnerability. One interviewee commented that "organizations are using just a single budget line or half an outcome around climate change and saving they are doing climate change adaptation (Interviewee 10, INGO representative)." Eriksen et al. (2021) also found a similar trend, which they referred to as 'retrofitting development as adaptation'. However, the results indicate that existing funding structures and mechanisms are also perpetuating this trend. The current funding mechanisms in the development sector are focused on competitive, projectbased approaches that neglect the need for contextual and need-based vulnerability and capacity assessments. Comprehensive assessments are typically time-consuming and resourceintensive, and organizations often do not see incentives to conduct them. Similarly, poor monitoring and post-evaluation practices are pervasive in the sector, as discussed in section 6.3.

Funding mechanisms have a crucial role to play in addressing these issues. With the notion of 'mainstreaming' as expressed by many practitioners, it is clear that the existing climate funds are simply funding development activities with certain climate aspects in it. Organizations continue to engage in such practices to secure funds and comply with finance criteria. There is empirical evidence and critical scholars have been raising the concern of 'rebranding' (Bertana et al., 2022; Eriksen et al., 2021; Schipper et al., 2020). Although it may not always lead to maladaptive pathways, it has clear implications in unequitable distribution of risk and benefits which is against the principle of effective adaptation. Effective adaptation requires patient, predictable, flexible, and easily accessible funding (Soanes et al., 2021). There was a strong consensus among interviewees regarding the need for localized financing, which aligns with innovative finance models such as The Devolved Climate Finance (IIED, 2021). The bureaucratic funding model, which relies on short-term project-based financing, traps projects

within unsustainable decision making structures (Coger et al., 2022). Current practice does not consider much nor provide adequate time to experiment, learn, and value failure.

6.5 Conceptual ambiguity

Despite the very marginal distinction between adaptation and development, aid agencies continue to separate them, and practitioners have expressed diverse opinions on the extent to which development should be a part of adaptation. As one interviewee proposed, "there is not a huge difference between development and adaptation as the agendas have strong links, it is just the way that adaptation implemented in the ground is very different (Interviewee 1, researcher)." Such an artificial distinction between development and adaptation has implications for how organizations adjust their strategies and projects to account for climate change concerns and secure funding (Bertana et al., 2022). Despite being an elusive concept, interviewee acknowledge that maladaptation has pushed practitioners towards realizing that these very positive adaptations also need to be examined holistically, for instance, changing the scale at which we are looking at issues. The critical adaptation literature also suggests that not every action reduces vulnerability (Atteridge & Remling, 2018; Eriksen et al., 2011). Similarly, the results show a significant difference in understanding what constitutes adaptation across a range of practitioners involved in the planning and implementation process. This is due to the temporal and spatial dimensions and the complexities associated with them. As noted by one interviewee, "we have a very poor sense of what a successful adaptation looks like, is it supposed to be successful in five years or is it supposed to be successful forever." However, these subjective aspects of (mal)adaptation are unavoidable and should not be dismissed. It is essential to bring multiple perspectives to the table and appreciate them accordingly. This means reframing and evaluating effectiveness situating them in terms of what happens in the everyday realities of the people a project works with. Considering the changing societal landscape, it is unfair to expect that an action or response against a particular risk at a particular time will always be effective (Dilling et al., 2015). Instead, the emphasis should be on adaptation as a dynamic and iterative process.

Given the definitional ambiguity of what constitutes adaptation, it is also difficult to depict adaptation actions from development measures (Singh et al., 2022). As a result, there is ongoing debate about the extent to which development should be a part of adaptation or vice versa, and how best to integrate them (Bertana et al., 2022). However, this trend shows that climate change and its consequences have shifted the focus internationally funded projects. Most projects labelled as adaptation are actually development measures, and organizations adjust their existing guidelines or project descriptions (Bertana et al., 2022). This 'retrofitting of adaptation' as Eriksen et al. (2021) terms it, is a contributing factors for pathways towards maladaptation. Although there could be various reasons for this, securing funding is key (section 6.4), and the funding stream demands to continue doing so. As expressed by some interviewees, there is always a sense of fear with the local NGOs that the donors could withdraw funding or affect future funding, if something did not go right instead of experimenting and learning. Besides, another problem identified is the way in which concepts

travel across the chain of stakeholders involved, from donors to a community-based organization. Organizations often take concepts and translate them along, which can result in a different meaning when it comes to the ground.

6.6 Culture of learning

The results show that one of the major drawbacks of adaptation practice is the failure to learn and adapt accordingly to make projects effective. Knowledge sharing is a key principle for successful adaptation; sharing information about what worked, what did not work, why it did not work, and how to improve in the future. It is equally important to learn from malpractice or failure because they can help future projects. The interviews revealed a strong emphasis on learning among the participants, with many expressing transparencies and promoting adaptive learning. For example, one interviewee said, "we need an honest overview or collection of failures so that we can analyze and learn from them (Interview 14, climate finance expert)." In contrast, a few participants held an opposing view and scepticism about its practical feasibility, citing "donors understand when the failures are due to external factors beyond the project scope and control, but when it comes to failures due to organizational issues, donors would not be happy to articulate that (Interviewee 15, INGO representative)." However, participants acknowledged that adaptation is the biggest learning cycle, and it is crucial to incorporate this learning into future project planning and design. Thus, there should be a feeding system where real-time information can be accessed so that other development agencies can find useful information before planning for investment in the area where other organizations have already worked. Therefore, learning should be performed within and across organizations. However, in current practice, much information and learning are held within a particular organization or limited network. Often, the project implementing partner collects and reports learning to the leading partner and donors along the chain, but they are unaware of how it is used in practice, as expressed by one partner NGO representative, "we never know if our learning has been translated in practice or not". So, learning and best practices are just one of many project documents and reporting formats that have no significance in planning more effective projects.

While projects proposals, guidelines, and frameworks simultaneously mention monitoring, evaluation, and learning, the learning process tends to be forgotten. A significant focus is on single-loop learning, which evaluates how well the planned activities are being implemented, rather than double loop learning which encompasses action learning interactions (Eriksen et al., 2021; Werners et al., 2021; Wojewska et al., 2021). One way to promote such a learning culture is to have the right cohort of organizations, trust, and partnerships. Having one table where the donor and implementer would sit together, discuss how far we have gone and what changes need to be done rather than imposing them with rigid logframes with sets of indicators to be reported at a certain time. This is different from the so called 'stakeholder meeting' or 'learning workshops' in the current practice; it should go far beyond and be much more transparent and honest (Soanes et al., 2021; Taylor et al., 2022). For this, mutual understanding, trust, and collective effort are required (Custer et al., 2022; Gajjar et al., 2022). It requires organizations that are not afraid to fail, donors to accept and admire that organizations are not

perfect and challenging themselves to deal with complex issues, and lastly, local honest partners who think beyond pleasing donors. One interviewee calls this a "*new way of working, collaborative approach of development partners, governments and INGOs which is not about competition and what visible impact a particular NGO has brought in a particular geographic area.*"

6.7 Shift from shallow understanding towards holistic perspective

Across the interviews, there was a notable demand for holistic, long-term strategic thinking and programming perspectives. In contrast, current practice seems to be limited within the project scope, and projects rarely consider the drivers of vulnerability. Interviewees' understanding of climate change adaptation clearly shows that it is predominantly coping and incremental (Figure 2), unfolding significantly through short-term technical interventions without recognizing how climate change impacts on people are deeply rooted within the system they are in. This is evident in the interview response and projects (5.2 and 5.3) emphasis on advanced irrigation systems, distribution of drought resistant seeds to support economic growth, rainwater harvesting techniques, boreholes and water collection ponds, etc. Apparently, all projects suggest that there is often a focus on biophysical interventions without considering system dynamics. Scientific literature also argues that such project interventions are driven by the concept of outcome vulnerability, in other words, the scientific framing of CCA (Nagoda, 2015). This does not necessarily mean that such an approach is wrong or maladaptive, yet they have proven to be ineffective in reducing vulnerability in the long term. Nagoda (2015) found that households with better adaptive capacity often mentioned limitations, such as drought, lack of technical support, irrigation and manure, as the causes of vulnerability. Whereas food insecure and poor households from the same community referred to social causes such as exclusion from decision-making process, limited access to land, water, and other resources, access to political networks and education.

Very few interviewees, especially independent researchers, reflected and questioned "are we actually addressing the problems that communities are facing or just addressing the impacts of climate change." There needs to be resourcing and thinking about building capacity within the community projects work with rather than leaving and taking away expertise once the project is completed. Interviewees expressed a preference for long-term programming, which is not always practiced. While it is not feasible for a single project to solve every issue, collective action and collaborative approaches are crucial to building multiple projects that support one another. In line with this argument, one interviewee shared that "sometimes NGOs design projects with a lot of components and rarely address each in-depth". So, having a holistic perspective does not mean including every issue in a project; rather taking a step back and viewing the challenge the project aims to address from multiple perspectives.

These findings are consistent with those reported by Bertana et al., (2022), that technological approaches and solutions dominate adaptation projects significantly. This views climate

change as a problem that can be addressed by technical scientific knowledge which confines its understanding and scope. Adaptation from this perspective is inferred as a means of protecting people and property from external threats, mainly climate-related events. Such a narrow conception leads to irreversible adaptation actions with a high risk of maladaptive outcomes (Barnett & O'Neill, 2010; Hallegatte, 2009). This tends to ignore the fact that all adaptation projects operate within, and are influenced by, social, political, cultural, and economic relations (Eriksen et al., 2015). This ultimately ignores the ontologies of climate change and the co-emergence of society and nature by presenting climate change as an external threat to both natural and human systems (Nightingale et al., 2020). The ontological perspective of the communities that the project target, should be at the centre of adaptation project design and implementation rather than solely focusing on technocratic claims of adaptation knowledge (Mills-Novoa, 2023). Technocratic and apolitical approaches to adaptation exclude the most marginalized from the processes that are intended for them and ultimately contribute to maintaining the status quo (Nagoda & Nightingale, 2017; Nightingale et al., 2022). Even if they are included as part of the formality, they do not have a meaningful influence on what decisions and policies are made. Case study projects are no exception to this, and they primarily propose technocratic and apolitical interventions targeting economic growth via increased agricultural productivity and better market opportunities, instead of measures to address the causes of inequality and vulnerability deeply rooted in society. Similarly, adaptation projects should not consider communities as homogenous entities and should consider the nuances in social hierarchies, decision-making processes, and social dynamics (Buggy & McNamara, 2016). Scholars also demand to reinterpret the sense of 'community' in community-based climate change adaptation. The community is beyond a mere place where projects are implemented (Buggy & McNamara, 2016). It should be conceptualized beyond the geographical boundary, and the underlying context should not be overlooked (section 6.1). Simply, "community" is often romanticized and assumed to be harmonious. Consequently, internationally funded projects tend to ignore the inherently complex, diverse, and multifaceted nature of the community.

As is evident in the literature, adaptation efforts should be rooted in a broader framing derived from an epistemologically plural approach to context and drivers (Ensor et al., 2019). This means understanding the experiences of communities through multiple lenses, such as climate risk management, social dynamics, economic changes and political contexts. Additionally, it is important to ask questions about the significant changes people's lives are experiencing due to climate change rather than just focusing on biophysical impacts alone. Finally, while planning projects, organizations need to draw upon different disciplines that can provide insights into viable solutions beyond those anticipated by specialist knowledge alone. To implement adaptation effectively, organizations should first understand what adaptation is, what/who is being adapted to, who implements adaptation, how adaptation occurs, and its outcomes (Asare-Nuamah et al., 2021). One of the approaches suggested by a key expert is to ensure that adaptive management is built within project design and to look at adaptation within a complex adaptive system (Figure 4). While it may seem too academic and idealistic, approaching with the idea that our systems are inherently dynamic and complex will challenge existing adaptation practices in many ways. One interviewee further emphasized that

"adaptation and adaptation management are lost", there is not much monitoring of how certain adaptation interventions are tied to people's well-being outcome. This again comes back to the framing of climate risk and how it is understood with a narrow focus. Maladaptation could be a great tool to help us recognize the need to change our understanding of risk. For instance, one interviewee suggested that "*if adaptation practitioners or organizations step back* from the narrow metrics such as amount of crop protected from adaptation interventions to people's wellbeing as an outcome, we will be able to identify several different interventions far broader than what is currently being considered."

6.8 Project problems

The majority of the interviewees expressed their concern about the complexity of adaptation, especially in terms of the potential impacts the projects could have. It is not always possible to foresee adaptation outcomes as several indirect factors could influence this. Given the uncertainty, practitioners frequently referred to a common concern of flexible programming, with one participant summarizing it as "flexibility ensures that the program can respond to unforeseen challenges or things that have not been considered". Emphasizing this, one of the local NGO representatives shared that, depending on the donors, sometimes it takes up more than 9-10 months to make some revisions to the project timelines. Therefore, flexible programming provides local actors the freedom to embrace trial and error and allows them to respond to dynamic and emergent risk factors. When programs have built-in mechanisms for adaptive management, implementing NGOs can accommodate change based on their learning of what works and what does not (Villanueva et al., 2018). For instance, project plans, TOCs, and indicators could be adapted as the local context demands. Despite good intentions and acknowledging the importance of community-level participation, project management, and planning ends up being predominantly top-down, leveraging the risk of maladaptive outcomes. Although current practice might seem participatory because of the consultation events that are often organized in different phases of projects, these events actually go through pre-existing governance institutions and power relations, thus excluding marginalized individuals. In other words, so-called "community-based participation" or "participatory processes" exclude the marginalized due to their inherent power relations, and ultimately generate projects with a high risk of exacerbating inequalities (section 6.1). Participation is not just 'stakeholder engagement', it is going beyond this and exploring differential interests, values, and knowledge production and contestation. It is moving away from the decision-making process where marginalized people do not have the opportunity to define problems and solutions. Effective adaptation entails challenging development paradigms that marginalize certain populations.

It is evident from the interviews that there is widespread consensus on the need for long-term programs. The short-term nature of projects is one of the reasons why much focus is on project outputs. In the words of an interviewee, "*NGOs are happy to see the number of beneficiaries reached, number of people participated in the training, but what happens after that (Interviewee 5, community-based NGO)*." In a short-term duration project, much M&E time is spent on administrative aspects, data collection, reporting, and less or no attention is paid to

the real impact of the project. Even if certain NGOs want to do, for instance post-project evaluation, there is no incentive because this is not built in, or the donors never want it. This is not a new finding in development studies and short-term projects have always been critiqued in terms of sustainable outcomes (Hagelsteen & Becker, 2019). Furthermore, some interviewee reported that governments often do not take ownership of these internationally funded projects, stating that "government line ministries are supposed to own these projects and make follow ups in monitoring...... unfortunately, this is not the case (Interviewee 11, INGO representative)." Projects proposal analysis revealed that the lack of ownership can be attributed to inadequate coordination during the project's preliminary design phase. Ownership needs to be developed throughout the project, as simply expecting governments to take over interventions that they never wanted is not a viable solution. The key question to be asked is, was the project intervention based on the needs of the communities? was it demanded by the local government? For example, the establishment of Farmer Producer Companies (FPCs) and their socioeconomic empowerment was one of the key project interventions of the women empowerment project in India (5.2). However, despite being facilitated by NGOs, it was quite challenging for the FPCs to be recognized by the government as eligible for financial support and grant for schemes, as shared by one community-based interviewee. This highlights the need for more coordinated and collaborative work during the project planning. Similarly, sustainability or exit strategy of the projects often do not function well because they do not acknowledge the long-term challenges faced by communities. A key fundamental question here is who should be responsible for sustaining the outcomes of the adaptation projects. Projects proposals demonstrates that the current sustainability mechanisms of the projects rely heavily on households, communities, local institutions, and governments with low capacity and their competing needs. Similarly, this does not acknowledge the future climate change impacts and shifting needs of communities on the frontline (Mills-Novoa, 2023). Such a reliance on local actors with limited resources and capacity needs to be replaced with concrete and funded sustainability mechanisms (Mills-Novoa, 2023). In any case, it is not only the project donor's and implementor's responsibility to ensure a sustainable outcome. Local actors indeed play a key role to create systems, structures, and policies. To achieve this, project sustainability mechanism in the first place should be designed in coordination with the frontline communities and individuals. In addition, donors need to explore the possibility of ex-post evaluation and ensure the provision of resources, funds, and technical support from the onset of projects.

7. Conclusion and Recommendations

As the climate finance mechanisms mature and climate change impacts become more frequent and intense, the scale and number of adaptation projects are expected to grow over the next decade. Vulnerable nations continue to depend upon intergovernmental funding and international non-governmental organizations for adaptation projects. Therefore, it is essential to understand the outcomes and underlying processes of such projects. Meanwhile, significant advancement in climate change discourse and critique of adaptation, has opened the maladaptation debate. This thesis aimed to explore practitioners' perspective on identifying and addressing the potential risk of maladaptation, guided by three key research questions:

(i) What are the principles that characterize effective adaptation?

The results showed that the practitioners' understanding of effective adaptation is essentially in accordance with the principles of locally-led adaptation (Soanes et al., 2021). Community based approach was found to be the key factor characterizing effective adaptation. However, despite the efforts and focus on community involvement, internationally funded projects often end up being predominantly top-down due to structural issues, such as requirements and standards, which prevent full-fledged community-led initiatives¹. Although the participatory approach is widely recognized in theory, the realities of the aid landscapes and development practices present significant barriers and hinder the creation of enabling conditions for its implementation (section 6.7 and 6.8). In addition, there is a significant difference in understanding of what successful adaptation entails, among the interviewees across different levels (Table 1). While organizations should not strive for a widely accepted or common definition of successful adaptation, there should at least be a shared understanding across organizations involved in the project. The results highlighted that 'justice and equity aspect' could be that common understanding point, and it should be at the centre of adaptation planning. To ensure and promote equitable, inclusive, and sustainable adaptation projects, it is recommended that international aid organizations adopt an adaptation process that is guided by the principles of locally led, justice focused and equitable adaptation. This would help the organization to ensure that the adaptation projects do not exacerbate the root causes of vulnerability.

(ii) What are the pre-determining factors that might lead to the negative consequences of adaptation projects?

Interviews and projects documents analysis revealed that failure to consider contextual factors and insufficient monitoring and evaluation are the major factors determining maladaptive pathways. Such contextual factors include power dynamics, varying vulnerability, competing needs, and behavioural/cultural aspects. Failure to consider these factors runs the risk of producing limited or no impact on reducing vulnerability of the communities' projects work with. In the worst case, it might result in adverse impacts undermining the resilience and adaptive capacity. In addition, context specific climate risk scenarios also play a vital role, which is not often acknowledged by the adaptation projects. Therefore, I/NGOs can incorporate comprehensive participatory and contextually driven analyses, such as socio-political, risk, and vulnerability analyses, to address this issue. This should prioritize the identification of contextual issues, enabling organizations to effectively tailor their projects to the communities they serve. Similarly, it was identified that limited monitoring and evaluation is one of the factors leading to negative outcomes of the adaptation projects. Narrow monitoring metrices and rigid frameworks do not allow practitioners to be flexible to the dynamic contexts project

¹ Community-based and Community-led approach mentioned here refers to slightly different approaches. In a community-based approach, local actors may be invited to plan and implement certain intervention, may be able to direct some priorities, and funding reaches the community via intermediate. Whereas, in community-led approach, local actors decide what is implemented, how and by whom. Likewise, local capacities are supported to design, implement, monitor, and maintain measures.

operates in. So, it is crucial for organizations to co-create such tools in coordination with the intended beneficiaries of the project. The implications of this study suggest that existing monitoring and evaluation tools should be adapted and equipped with essential time and resources, thinking beyond the project timeline. Organizations should consider the long-term risks and outcomes associated with the projects and build them into the project frameworks.

(iii) What are the challenges in the existing practice of planning and implementing adaptation projects for I/NGOs? How can these be addressed?

This research identified structural issues in funding mechanism, conceptual ambiguity, absence of learning culture, narrow perspective, and poor project practice as key challenges in terms of planning and implementing effective adaptation projects. The current aid mechanisms have structural issues preventing them from reaching the grassroots level and prioritizing the needs of those most affected by climate change. Short-term, project-based financing creates unsustainable decision-making structures, disproportionately benefiting certain groups, leading to wider inequalities and vulnerabilities. Therefore, aid agencies should promote patient, predictable and flexible funding, making it easier for community-based organizations to access finance, and allowing communities to make adaptation decisions based on their needs. Moreover, this also leads to conceptual ambiguities where organizations adjust existing projects as adaptation to secure funding. This research emphasizes the need for a transparent and collaborative approach among organizations and donors to promote a learning culture and incorporate it into future project planning. Lastly, the research argues that current practices focus too narrowly on short-term technical interventions without considering the root causes of vulnerability. Given the results of this research, it is recommended that organizations should promote long-term flexible programming and a collaborative approach that considers multiple perspectives, acknowledging the dynamic context of climate risk.

While this finding is not new, it takes the discussion further and raises concerns among aid organizations to rethink how they are approaching adaptation. In conclusion, this thesis has contributed to the burgeoning literature on maladaptation from practitioners' perspectives, shedding light on the challenges and opportunities. There is no easy practice for transformative adaptation and no silver bullet to ensure projects do not have maladaptive outcomes. It is imperative for donors and implementing organizations to acknowledge this and shift the narrow perspective, decision-making, governance, and institutional culture. Overall, it is recommended that future research explores maladaptation concept as a tool to reshape current adaptation practices among organizations and ensure effective and sustainable adaptation. Additionally, further research is needed to understand how donor and partner organization relations shape decision-making processes and effectiveness of the project outcomes.

The following table provides guiding questions for the practitioners to be more reflexive and consider potential negative outcomes of the projects. As reported by the interviewees and noticed during the research, there is often not much conversation among organizations and stakeholders regarding the potential negative consequences of adaptation actions. So, the table suggest some questions that could be applied and reflected upon during different phases of

project management. This aims to raise concerns among practitioners and foster learning from each other's experiences. It is important to note that the purpose of the checklist is not to identify maladaptive outcome in specific, rather to ensure, right from the beginning, that factors outlined in Section 6 are not overlooked. These factors may inadvertently lead to unintended and unforeseen effects. The checklist serves as a discussion tool, allowing practitioners to critically assess their planning and implementation of projects, highlighting areas that may require improvement.

Table 3: Recommendation for organizations to address the existing gaps and challenges identified in adaptation practice. The guiding questions are structured around key themes identified by the research. To simplify, the more the questions of the checklist are answered with 'no' or 'partially', the higher the maladaptation risk. Inspired from (REGILIENCE consortium, 2023) self-assessment tool on maladaptation.

Key factors				
identified	Guiding question to reflect upon the current adaptation practice	Yes	Partially	No
Contextual understanding	Is the project considering current and future climate risk scenarios of the project target region? Given the dynamic nature and context, existing risk could be intensified by climate variability or new risk could emerge in the worst case. Thus, adaptation projects should not only consider current scenario, but expected future climatic risks (Interviews)			
	Is the inherent political/economic power dynamics explicitly acknowledged by the project?			
	Vested interest of certain groups, trade offs between whose interests are being served and neglected? Who decides and who is being excluded from the decision making process? Project should be explicit on why adaptation priorities have been in certain way, the role of instituion and governance. (Interviews, Glover & Granberg, 2020; Nightingale et al., 2020)			
	Are the project interventions tailored to contextual needs, priorities, behaviorial, cultural and other contextual factors?			
	from the reality (interviews)			
Justice or 4	Does the project consider the root causes of marginality and differential vulnerability? Project should understand the distribution of risk and benefit so that it would avoid reinventing the existing disparities (Interviews)			
Equity	Dess the project consider different consists of justice?			
24.00	Who suffers from climate risk and who benefits from adaptation response? Who makes adaptation decisions? And whose concerns matter? (Adger et al., 2005; Singh et al., 2022, Taylor et al., 2022)			

	Is the project equipped with monitoring mechanism to interrogate any possible adverse or unintended outcome?	
Inadequate monitoring and	Monitoring instrument such as risk matrix should also focus on potential risk beyond the project timeline, rigid monitoring tools heavily focused on planned activities vs acheivement should be adapted. (Project Document Analysis and Interviews)	
evaluation	Is the post project evaluation built in the project structure?	
	Despite the methodological and conceptual challenges, it is important to assess and evaluate both effective and negative impacts of projects once the project is completed. (Interviews)	
	Does the funding model allocate time and resources for conextual vulnerability and capacity assessment?	
Funding structure and mechanism	If not, communication, open dialogue, and common understanding between donors, and implementors is key. What role can a organization play to ensure this? Limited interview with donors and guidelines/principles endorsed indicates that, everyone accept the need to shift adaptation practice. Using maladaptation, as a dicussion tool, organizations should take the agenda further	
	Is the finance reaching to the people/communities in real need?	
	Or is it simply following the pre-existing network and structure which have been the historical cause of marginalisation. Organizations should reflect and be explicit (K. Browne & Claudien Razafiarimanana, 2022)	
	Is the project or organization explicit on how their framing influences adaptation practice?	
Conceptual ambiguity	Project should constitute platform for engaging diverse knowledge beyond the limited scientific and expert knowledge.	
anno igunoj	Is the project reflexive on adaptation framing and decisions?	
	Organnizations should critically reflect upon who is currently and who should be defining 'good' adaptation (Piggott-McKellar, Nunn, et al., 2020))	
	Does the project offer space for experimenting and learning?	
Learning culture	Considering the uncertianty and system dynamics, organization should accept that projects might fail (failed project does not necessarily mean maladaptation, figure 3), the key priority should be to learn and adapt	
	Does the project acknowledge adaptive management?	
	Learning by the project stakeholders should be translated in practice instead of collecting it just for the sake of reporting and presenting (Interviews)	
	Is the learning held within the organization? Is it helpful for future projects?	
	There should be a common platform where organizations working on climate change adaptation can learn from each other. Organizations should adapt themselves first before helping communities to adapt (Interviews)	

	Is the project focused on outcome vulnerability or contextual vulnerability?		
Shift towards	Vulnerability is not just the outcomes of climate change impacts, but a complex interaction of pre existing conditions that make certain people more vulnerable than others (O'Brien et al., 2007)		
perspective	Is the project overemphasizing on scientific or technocentric framing of adaptation?		
	Adaptation operates within contextual factors and thus project should think beyond technical fixes (Bertana et al., 2022)		
	Is the project flexible enough to respond to uncertainty and climatic variability?		
Project nature and structure	Rigid programme structure does not adress the needs and priorities and leads to failed adaptation. This is also not consistent with the dynamic nature of risk and vulnerability (Interviews)		
	Does the project commit resources to invest in local instituional strengthening, partnership and relationship building rather than mere stakeholder consultation?		
	This is one of the principles of locally led adaptation to ensure sustained local leadership and enhanced adaptive capacity. So, projects should be explicit on this. (Coger et al., 2022; Taylor et al., 2022)		
	Is the organization taking programmatic and portfolio appraoches or just pilot projects without explicit sustainability and learning mechanism?		
	Programmatic approach provides opportunity and time for re-evaluating outcomes and impacts, enhance trust and relationship (Interviews, Taylor et al., 2022)		

8. References

- Acharya, S., Pathak, K., & Pandey, C. L. (2021). Does High-Value Agriculture Build Resilient Livelihood? Insights from the Study of Cardamom and Vegetable Farmers of Nepal. *Journal of Forest and Livelihood*.
- Adam, H. N., Kjosavik, D. J., & Shanmugaratnam, N. (2018). Adaptation trajectories and challenges in the Western Ghats: A case study of Attappady, south India. *Journal of Rural Studies*, *61*, 1–11. Scopus. https://doi.org/10.1016/j.jrurstud.2018.05.002
- Adger, W. N. (2006). Vulnerability. *Global Environmental Change*, *16*(3), 268–281. https://doi.org/10.1016/j.gloenvcha.2006.02.006
- Adger, W. N., Arnell, N. W., & Tompkins, E. L. (2005). Successful adaptation to climate change across scales. *Global Environmental Change-Human and Policy Dimensions*, *15*(2), 77–86. https://doi.org/10.1016/j.gloenvcha.2004.12.005
- Adger, W. N., Dessai, S., Goulden, M., Hulme, M., Lorenzoni, I., Nelson, D. R., Naess, L. O., Wolf, J., & Wreford, A. (2009). Are there social limits to adaptation to climate change? *Climatic Change*, 93(3–4), 335–354. https://doi.org/10.1007/s10584-008-9520-z
- Antwi-Agyei, P., Dougill, A. J., Stringer, L. C., & Codjoe, S. N. A. (2018). Adaptation opportunities and maladaptive outcomes in climate vulnerability hotspots of northern Ghana. *Climate Risk Management*, 19, 83–93. https://doi.org/10.1016/j.crm.2017.11.003
- Asare-Nuamah, P., Dick-Sagoe, C., & Ayivor, R. (2021). Farmers' maladaptation: Eroding sustainable development, rebounding and shifting vulnerability in smallholder agriculture system. *Environmental Development, null,* null. https://doi.org/10.1016/j.envdev.2021.100680
- Atteridge, A., & Remling, E. (2018). Is adaptation reducing vulnerability or redistributing it? *WIREs Climate Change*, *9*(1). https://doi.org/10.1002/wcc.500
- Ayanlade, A., Smucker, T. A., Nyasimi, M., Sterly, H., Weldemariam, L. F., & Simpson, N. P. (2023). Complex climate change risk and emerging directions for vulnerability research in Africa. *Climate Risk Management*, 40, 100497. https://doi.org/10.1016/j.crm.2023.100497
- Barnett, J., Mortreux, C., & Adger, W. N. (2013). Barriers and limits to adaptation: Cautionary notes. In *Natural Disasters and Adaptation to Climate Change* (Vol. 9781107010161, pp. 223–235). Scopus. https://doi.org/10.1017/CBO9780511845710.029
- Barnett, J., & O'Neill, S. (2010). Maladaptation. *Global Environmental Change*, 20(2), 211–213. https://doi.org/10.1016/j.gloenvcha.2009.11.004

Bassett, T. J., & Fogelman, C. (2013). Déjà vu or something new? The adaptation concept in the climate change literature. *Geoforum*, *48*, 42–53. https://doi.org/10.1016/j.geoforum.2013.04.010

Bazeley, P., & Jackson, K. (2007). *Qualitative Data Analysis with NVivo*.

- Berrang-Ford, L., Siders, A. R., Lesnikowski, A., Fischer, A., Callaghan, M., Haddaway, N., Mach, K., Araos, M., Shah, M. A. R., Wannewitz, M., Doshi, D., Leiter, T., Matavel, C., Musah-Surugu, I., Wong-Parodi, G., Antwi-Agyei, P., Ajibade, I., Chauhan, N., Kakenmaster, W., & Minx, J. (2021). A systematic global stocktake of evidence on human adaptation to climate change. *Nature Climate Change*, *11*. https://doi.org/10.1038/s41558-021-01170-y
- Bertana, A., Clark, B., Benney, T. M., & Quackenbush, C. (2022). Beyond maladaptation: Structural barriers to successful adaptation. *Environmental Sociology*, 8(4), 448–458. https://doi.org/10.1080/23251042.2022.2068224
- Blaikie, P., Wisner, B., Cannon, T., & Davis, I. (1994). *At Risk: Natural Hazards, People's Vulnerability and Disasters*. Routledge. https://doi.org/10.4324/9780203974575
- Bours, D., McGinn, C., & Pringle, P. (2014). *Monitoring & evaluation for climate change* adaptation and resilience: A synthesis of tools, frameworks and approaches - Second edition. https://doi.org/10.13140/RG.2.1.1151.4645
- Boutroue, B., Bourblanc, M., Mayaux, P.-L., Ghiotti, S., & Hrabanski, M. (2022). The politics of defining maladaptation: Enduring contestations over three (mal)adaptive water projects in France, Spain and South Africa. *International Journal of Agricultural Sustainability*, 20(5), 892–910. https://doi.org/10.1080/14735903.2021.2015085
- Brink, E., Falla, A. M. V., & Boyd, E. (2023). Weapons of the vulnerable? A review of popular resistance to climate adaptation. *Global Environmental Change*, *80*, 102656. https://doi.org/10.1016/j.gloenvcha.2023.102656
- Brown, K. (2011). Sustainable adaptation: An oxymoron? *Climate and Development*, *3*(1), 21–31. https://doi.org/10.3763/cdev.2010.0062
- Brown, K., Naylor, L., & Quinn, T. (2017). Making Space for Proactive Adaptation of Rapidly Changing Coasts: A Windows of Opportunity Approach. *Sustainability*, *9*(8), 1408. https://doi.org/10.3390/su9081408
- Browne, K., & Razafiarimanana, C. (2022). Adaptation finance failing to reach the most vulnerable: A multi-level model of household political power in Madagascar. *PLOS Climate*. https://doi.org/10.1371/journal.pclm.0000050
- Buggy, L., & McNamara, K. E. (2016). The need to reinterpret "community" for climate change adaptation: A case study of Pele Island, Vanuatu. *Climate and Development*, 8(3), 270–280. Scopus. https://doi.org/10.1080/17565529.2015.1041445

- Câmpeanu, C. N., & Fazey, I. (2014). Adaptation and pathways of change and response: A case study from Eastern Europe. *Global Environmental Change*, *28*, 351–367. https://doi.org/10.1016/j.gloenvcha.2014.04.010
- Canales, N., Klein, R. J. T., Bakhtaoui, I., & Macura, B. (2023). Assessing adaptation progress for the global stocktake. *Nature Climate Change*, 1–2. https://doi.org/10.1038/s41558-023-01656-x
- Charmaz, K. (2006). Constructing Grounded Theory: A Practical Guide through Qualitative Analysis. SAGE Publications. http://ebookcentral.proquest.com/lib/lund/detail.action?docID=585415
- Chi, C.-F., Lu, S.-Y., Hallgren, W., Ware, D., & Tomlinson, R. (2021). Role of Spatial Analysis in Avoiding Climate Change Maladaptation: A Systematic Review. *Sustainability*, 13(6), 3450. https://doi.org/10.3390/su13063450
- Chi, C.-F., Lu, S.-Y., & Lee, J.-D. (2020). Ostensibly Effective Adaptive Measures Could Potentially Be Maladaptations: A Case Study of the Jiadung Coastal Area, Pingtung County, Taiwan. *Coastal Management*, 48(6), 643–676. https://doi.org/10.1080/08920753.2020.1803575
- Chmutina, K., von Meding, J., Williams, D. A., Vickery, J., & Purdum, C. (2022). From pity to fear: Security as a mechanism for (re)production of vulnerability. *Disasters*, disa.12568. https://doi.org/10.1111/disa.12568
- Coger, T., Corry, S., & Gregorowski, R. (2021). Reshaping Monitoring, Evaluation and Learning for Locally Led Adaptation. *World Resources Institute*. https://doi.org/10.46830/wriwp.20.00060
- Coger, T., Dinshaw, A., Tye, S., Kratzer, B., Thazin Aung, M., Cunningham, E., Ramkissoon, C., Gupta, S., Bodrud-Doza, Md., Karamallis, A., Mbewe, S., Granderson, A., Dolcemascolo, G., Tewary, A., Mirza, A., & Carthy, A. (2022). Locally Led Adaptation: From Principles to Practice. *World Resources Institute*. https://doi.org/10.46830/wriwp.21.00142
- Creswell, J. W. (2013). *Qualitative inquiry and research design: Choosing among five approaches* (3rd ed). SAGE Publications.
- Custer, S., Horigoshi, A., Hutchinson, A., Choo, V., & Marshall, K. (2022). AidData / Aid Reimagined: How can foreign assistance support locally-led development? [Policy Report]. AidData, William & Mary's Global Research Institute. https://www.aiddata.org/publications/aid-reimagined-how-can-foreign-assistancesupport-locally-led-development
- Dilling, L., Daly, M. E., Travis, W. R., Wilhelmi, O. V., & Klein, R. A. (2015). The dynamics of vulnerability: Why adapting to climate variability will not always prepare us for

climate change: Dynamics of vulnerability. *Wiley Interdisciplinary Reviews: Climate Change*, *6*(4), 413–425. https://doi.org/10.1002/wcc.341

- Dolšak, N., & Prakash, A. (2018). *The politics of climate change adaptation* (Vol. 43, p. 341). Scopus. https://doi.org/10.1146/annurev-environ-102017-025739
- Ensor, J. E., Wennström, P., Bhatterai, A., Nightingale, A. J., Eriksen, S., & Sillmann, J. (2019). Asking the right questions in adaptation research and practice: Seeing beyond climate impacts in rural Nepal. *Environmental Science & Policy*, 94, 227–236. https://doi.org/10.1016/j.envsci.2019.01.013
- Eriksen, S., Aldunce, P., Bahinipati, C. S., Martins, R. D., Molefe, J. I., Nhemachena, C.,
 O'Brien, K., Olorunfemi, F., Park, J., Sygna, L., & Ulsrud, K. (2011). When not every response to climate change is a good one: Identifying principles for sustainable adaptation. *Climate and Development*, 3(1), 7–20. https://doi.org/10.3763/cdev.2010.0060
- Eriksen, S., & Brown, K. (2011). Sustainable adaptation to climate change. *Climate and Development*, *3*(1), 3–6. https://doi.org/10.3763/cdev.2010.0064
- Eriksen, S., Nightingale, A. J., & Eakin, H. (2015). Reframing adaptation: The political nature of climate change adaptation. *Global Environmental Change*, *35*, 523–533. https://doi.org/10.1016/j.gloenvcha.2015.09.014
- Eriksen, S., Schipper, E. L. F., Scoville-Simonds, M., Vincent, K., Adam, H. N., Brooks, N., Harding, B., Khatri, D., Lenaerts, L., Liverman, D., Mills-Novoa, M., Mosberg, M., Movik, S., Muok, B., Nightingale, A., Ojha, H., Sygna, L., Taylor, M., Vogel, C., & West, J. J. (2021). Adaptation interventions and their effect on vulnerability in developing countries: Help, hindrance or irrelevance? *World Development*, *141*, 105383. https://doi.org/10.1016/j.worlddev.2020.105383
- European Parliament. (2012). Climate change financing :the concept of additionality in the light of the commission proposal for a Development Cooperation Instrument (DCI) for 2014 2020. Publications Office. https://data.europa.eu/doi/10.2861/89763
- Fenton, A., Paavola, J., & Tallontire, A. (2017). The Role of Microfinance in Household Livelihood Adaptation in Satkhira District, Southwest Bangladesh. World Development, 92, 192–202. https://doi.org/10.1016/j.worlddev.2016.12.004
- Flick, U. (2018). *The SAGE Handbook of Qualitative Data Collection*. SAGE Publications Ltd. https://doi.org/10.4135/9781526416070
- Forsyth, T., & McDermott, C. L. (2022). When climate justice goes wrong: Maladaptation and deep co-production in transformative environmental science and policy. *Political Geography*, 98, 102691. https://doi.org/10.1016/j.polgeo.2022.102691

- Fünfgeld, H., & Schmid, B. (2020). Justice in climate change adaptation planning: Conceptual perspectives on emergent praxis. *Geographica Helvetica*, 75(4), 437–449. Scopus. https://doi.org/10.5194/gh-75-437-2020
- Füssel, H.-M. (2007). Vulnerability: A generally applicable conceptual framework for climate change research. *Global Environmental Change*, 17(2), 155–167. https://doi.org/10.1016/j.gloenvcha.2006.05.002
- Gajjar, S. P., Hofemeier, A., Rokitzki, M., & Richards, C. (2022). GOOD PRACTICES FOR ADAPTATION ACTION RESEARCH.
- Gaworek-Michalczenia, M. F., Sallu, S. M., Di Gregorio, M., Doggart, N., & Mbogo, J. (2022).
 Evaluating the impact of adaptation interventions on vulnerability and livelihood resilience. *Climate and Development*, *14*(10), 867–883. Scopus.
 https://doi.org/10.1080/17565529.2021.2018987
- GIZ. (2014). Monitoring and Evaluating Adaptation at Aggregated Levels: A Comparative Analysis of Ten Systems. International Institute for Sustainable Development. https://www.iisd.org/publications/report/monitoring-and-evaluating-adaptationaggregated-levels-comparative-analysis-ten
- Global Center on Adaptation. (2022). *State and Trends in Adaptation Report; Adaptation at the core of a prosperous Africa in an uncertain and warming world*.
- Glover, L., & Granberg, M. (2020). *The Politics of Adapting to Climate Change*. Springer International Publishing. https://doi.org/10.1007/978-3-030-46205-5
- Glover, L., & Granberg, M. (2021). The Politics of Maladaptation. *Climate*, *9*, 69. https://doi.org/10.3390/cli9050069
- Grbich, C. (2013). *Qualitative Data Analysis: An Introduction*. SAGE Publications Ltd. https://doi.org/10.4135/9781529799606
- Green Climate Fund. (2021). *Thematic brief: Enhancing integrity to avoid maladaptation*. Independent Integrity Unit, Green Climate Fund. https://iiu.greenclimate.fund/document/thematic-brief-enhancing-integrity-avoidmaladaptation
- Guodaar, L., Asante, F., Eshun, G., Abass, K., Afriyie, K., Appiah, D. O., Gyasi, R., Atampugre, G., Addai, P., & Kpenekuu, F. (2020). How do climate change adaptation strategies result in unintended maladaptive outcomes? Perspectives of tomato farmers. *International Journal of Vegetable Science*, *26*(1), 15–31. Scopus. https://doi.org/10.1080/19315260.2019.1573393
- Hagelsteen, M., & Becker, P. (2019). Systemic problems of capacity development for disaster risk reduction in a complex, uncertain, dynamic, and ambiguous world.

International Journal of Disaster Risk Reduction, 36, 101102. https://doi.org/10.1016/j.ijdrr.2019.101102

- Hallegatte, S. (2009). Strategies to adapt to an uncertain climate change. *Global Environmental Change*, *19*(2), 240–247. https://doi.org/10.1016/j.gloenvcha.2008.12.003
- Hallegatte, S., Rentschler, J., & Rozenberg, J. (2020). Adaptation Principles: A Guide for Designing Strategies for Climate Change Adaptation and Resilience. World Bank Group. http://hdl.handle.net/10986/34780
- Holland, E., Patel, S., Roe, D., & Shakya, C. (2022). Money where it matters for people, nature and climate: Driving change through support for local level decision making over resources and finance. International Institute for Environment and Development; IIED. https://www.iied.org/20966iied
- IIED. (2021). Devolved Climate Finance (DCF) Alliance. International Institute for Environment and Development. https://www.iied.org/devolved-climate-finance-dcfalliance
- Inderberg, T. H., Eriksen, S., O'Brien, K., & Sygna, L. (Eds.). (2014). *Climate Change Adaptation and Development* (0 ed.). Routledge. https://doi.org/10.4324/9781315774657
- IPCC. (2001). *Climate change 2001: Synthesis report* (R. T. Watson & D. L. Albritton, Eds.). Cambridge University Press. https://www.ipcc.ch/site/assets/uploads/2018/03/front-1.pdf
- IPCC. (2022). Climate Change 2022: Impacts, Adaptation and Vulnerability. Contribution of Working Group II to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. (H.-O. Pörtner, D. C. Roberts, M. M. B. Tignor, E. S. Poloczanska, K. Mintenbeck, A. Alegría, M. Craig, S. Langsdorf, S. Löschke, V. Möller, A. Okem, & B. Rama, Eds.). https://www.ipcc.ch/report/sixth-assessment-report-working-group-ii/
- Jones, L., Carabine, E., & Schipper, L. (2015). (Re)Conceptualising Maladaptation in Policy and Practice: Towards an Evaluative Framework. *SSRN Electronic Journal*. https://doi.org/10.2139/ssrn.2643009
- Juhola, S., Glaas, E., Linnér, B.-O., & Neset, T.-S. (2016). Redefining maladaptation. *Environmental Science & Policy*, 55, 135–140. https://doi.org/10.1016/j.envsci.2015.09.014
- Juhola, S., Käyhkö, J., Heikkinen, M., Neset, T.-S., & Tuhkanen, H. (2022). Governance and everyday adaptations? Examining the disconnect between planned and autonomous adaptation through justice outcomes. *Ecology and Society*, 27(4), art41. https://doi.org/10.5751/ES-13664-270441

- Kates, R. W., Travis, W. R., & Wilbanks, T. J. (2012). Transformational adaptation when incremental adaptations to climate change are insufficient. *Proceedings of the National Academy of Sciences of the United States of America*, 109(19), 7156–7161. https://doi.org/10.1073/pnas.1115521109
- Klepp, S., & Chavez-Rodriguez, L. (2018). A Critical Approach to Climate Change Adaptation: Discourses, Policies, and Practices. https://doi.org/10.4324/9781315165448
- Lama, P. D., Becker, P., & Bergström, J. (2017). Scrutinizing the relationship between adaptation and resilience: Longitudinal comparative case studies across shocks in two Nepalese villages. *International Journal of Disaster Risk Reduction*, 23, 193–203. https://doi.org/10.1016/j.ijdrr.2017.04.010
- Loginova, J., & Batterbury, S. P. J. (2019). Incremental, transitional and transformational adaptation to climate change in resource extraction regions. *Global Sustainability*, *2*, e17. https://doi.org/10.1017/sus.2019.14
- Ma, A. T. H., Wong, G. K. L., Cheung, L. T. O., Lo, A. Y., & Jim, C. Y. (2021). Climate change perception and adaptation of residents in Hong Kong. *Journal of Cleaner Production*, *288*. Scopus. https://doi.org/10.1016/j.jclepro.2020.125123
- Magnan, A. K. (2014). Avoiding maladaptation to climate change: Towards guiding principles.
- Magnan, A. K., E. Lisa F. Schipper, Schipper, E. L. F., & Duvat, V. (2020). Frontiers in Climate Change Adaptation Science: Advancing Guidelines to Design Adaptation Pathways. *Current Climate Change Reports*, 6(4), 166–177. https://doi.org/10.1007/s40641-020-00166-8
- Magnan, A. K., Schipper, E. L. F., Burkett, M., Bharwani, S., Burton, I., Eriksen, S., Gemenne, F., Schaar, J., & Ziervogel, G. (2016). Addressing the risk of maladaptation to climate change. WIREs Climate Change, 7(5), 646–665. https://doi.org/10.1002/wcc.409
- Mersha, A. A., & van Laerhoven, F. (2018). The interplay between planned and autonomous adaptation in response to climate change: Insights from rural Ethiopia. *World Development*, *107*, 87–97. https://doi.org/10.1016/j.worlddev.2018.03.001
- Mills-Novoa, M. (2023). What happens after climate change adaptation projects end: A community-based approach to ex-post assessment of adaptation projects. *Global Environmental Change*, 80, 102655. https://doi.org/10.1016/j.gloenvcha.2023.102655
- Moser, S. C., & Ekstrom, J. A. (2010). A framework to diagnose barriers to climate change adaptation. *Proceedings of the National Academy of Sciences*, *107*(51), 22026–22031. https://doi.org/10.1073/pnas.1007887107

- Mycoo, M. A. (2014). Autonomous household responses and urban governance capacity building for climate change adaptation: Georgetown, Guyana. *Urban Climate*, *9*, 134–154. https://doi.org/10.1016/j.uclim.2014.07.009
- Nagoda, S. (2015). New discourses but same old development approaches? Climate change adaptation policies, chronic food insecurity and development interventions in northwestern Nepal. *Global Environmental Change*, *35*, 570–579. https://doi.org/10.1016/j.gloenvcha.2015.08.014
- Nagoda, S., & Nightingale, A. J. (2017). Participation and Power in Climate Change Adaptation Policies: Vulnerability in Food Security Programs in Nepal. *World Development, 100,* 85–93. https://doi.org/10.1016/j.worlddev.2017.07.022
- Nick Brooks, S. F. (2014). *Tracking Adaptation and Measuring Development: A step-by-step guide*. International Institute for Environment and Development. https://www.iied.org/10100iied
- Nightingale, A. J. (2017). Power and politics in climate change adaptation efforts: Struggles over authority and recognition in the context of political instability. *Geoforum*, *84*, 11–20. https://doi.org/10.1016/j.geoforum.2017.05.011
- Nightingale, A. J., Eriksen, S., Taylor, M., Forsyth, T., Pelling, M., Newsham, A., Boyd, E., Brown, K., Harvey, B., Jones, L., Bezner Kerr, R., Mehta, L., Naess, L. O., Ockwell, D., Scoones, I., Tanner, T., & Whitfield, S. (2020). Beyond Technical Fixes: Climate solutions and the great derangement. *Climate and Development*, *12*(4), 343–352. https://doi.org/10.1080/17565529.2019.1624495
- Nightingale, A. J., Gonda, N., & Eriksen, S. H. (2022). Affective adaptation = effective transformation? Shifting the politics of climate change adaptation and transformation from the status quo. *WIREs Climate Change*, *13*(1). https://doi.org/10.1002/wcc.740
- Nunn, P. D., McLean, R., Dean, A., Fong, T., Iese, V., Katonivualiku, M., Klöck, C., Korovulavula, I., Kumar, R., & Tabe, T. (2020). Adaptation to climate change: Contemporary challenges and perspectives (p. 524). Scopus. https://doi.org/10.1007/978-3-030-32878-8_14
- O'Brien, K., Eriksen, S., Nygaard, L. P., & Schjolden, A. (2007). Why different interpretations of vulnerability matter in climate change discourses. *Climate Policy*, 7(1), 73–88. https://doi.org/10.1080/14693062.2007.9685639
- OECD. (2009). Integrating Climate Change Adaptation into Development Co-operation: Policy Guidance. OECD. https://doi.org/10.1787/9789264054950-en
- Ojha, H. R., Ghimire, S., Pain, A., Nightingale, A., Khatri, D. B., & Dhungana, H. (2016). Policy without politics: Technocratic control of climate change adaptation policy making in

Nepal. *Climate Policy*, *16*(4), 415–433. https://doi.org/10.1080/14693062.2014.1003775

- Owusu-Daaku, K. N. (2018). (Mal)Adaptation opportunism: When other interests take over stated or intended climate change adaptation objectives (and their unintended effects). *Local Environment*, *23*, 934–951. https://doi.org/10.1080/13549839.2018.1505836
- Park, S. E., Marshall, N. A., Jakku, E., Dowd, A. M., Howden, S. M., Mendham, E., & Fleming, A. (2012). Informing adaptation responses to climate change through theories of transformation. *Global Environmental Change*, 22(1), 115–126. https://doi.org/10.1016/j.gloenvcha.2011.10.003
- Pelling, M., O'Brien, K., & Matyas, D. (2015). Adaptation and transformation. *Climatic Change*, 133(1), 113–127. https://doi.org/10.1007/s10584-014-1303-0
- Piggott-McKellar, A. E., Nunn, P. D., McNamara, K. E., & Sekinini, S. T. (2020). Dam(n) Seawalls: A Case of Climate Change Maladaptation in Fiji. 69–84. https://doi.org/10.1007/978-3-030-40552-6_4
- Pritchard, B., & Thielemans, R. (2014). 'Rising Waters Don't Lift All Boats': A sustainable livelihood analysis of recursive cycles of vulnerability and maladaptation to flood risk in rural Bihar, India. *Australian Geographer*, 45(3), 325–339. https://doi.org/10.1080/00049182.2014.930001
- Quealy, H. M., & Yates, J. S. (2021). Situated adaptation: Tackling the production of vulnerability through transformative action in Sri Lanka's Dry Zone. *Global Environmental Change*, 71, 102374. https://doi.org/10.1016/j.gloenvcha.2021.102374
- Rahman, H. M. T., & Hickey, G. M. (2019). What Does Autonomous Adaptation to Climate Change Have to Teach Public Policy and Planning About Avoiding the Risks of Maladaptation in Bangladesh? *Frontiers in Environmental Science*, 7, 2. https://doi.org/10.3389/fenvs.2019.00002
- REGILIENCE consortium. (2023). *Self-assessment Tool for Maladaptation—Regilience*. Regional Pathways to Climate Resilience. https://regilience.eu/self-assessment-toolfor-maladaptation/
- Scheraga, J. D., & Grambsch, A. (1998). Risks, opportunities, and adaptation to climate change. *Climate Research*, *11*(1), 85–95. https://doi.org/10.3354/cr011085
- Schipper, E. L. F. (2020). Maladaptation: When Adaptation to Climate Change Goes Very Wrong. *One Earth*, *3*(4), 409–414. https://doi.org/10.1016/j.oneear.2020.09.014
- Schipper, E. L. F. (2022). Catching maladaptation before it happens. *Nature Climate Change*, *12*(7), 617–618.

- Schipper, E. L. F., Tanner, T., Dube, O. P., Adams, K. M., & Huq, S. (2020). The debate: Is global development adapting to climate change? *World Development Perspectives*, 18, 100205. https://doi.org/10.1016/j.wdp.2020.100205
- Sen, R. (2023). Salt in the wound: Embodied everyday adaptations to salinity intrusion in the Sundarbans. *Ecology and Society*, 28(2), art10. https://doi.org/10.5751/ES-14037-280210
- Simon, K., Diprose, G., & Thomas, A. C. (2020). Community-led initiatives for climate adaptation and mitigation. *Kōtuitui: New Zealand Journal of Social Sciences Online*, *15*(1), 93–105. https://doi.org/10.1080/1177083X.2019.1652659
- Simpson, N. P., Williams, P. A., Mach, K. J., Berrang-Ford, L., Biesbroek, R., Haasnoot, M., Segnon, A. C., Campbell, D., Musah-Surugu, J. I., Joe, E. T., Nunbogu, A. M., Sabour, S., Meyer, A. L. S., Andrews, T. M., Singh, C., Siders, A. R., Lawrence, J., van Aalst, M., & Trisos, C. H. (2023). Adaptation to compound climate risks: A systematic global stocktake. *IScience*, *26*(2), 105926. https://doi.org/10.1016/j.isci.2023.105926
- Singh, C., Iyer, S., New, M. G., Few, R., Kuchimanchi, B., Segnon, A. C., & Morchain, D. (2022). Interrogating 'effectiveness' in climate change adaptation: 11 guiding principles for adaptation research and practice. *Climate and Development*, 14(7), 650–664. https://doi.org/10.1080/17565529.2021.1964937
- Soanes, M., Bahadur, A., Shakya, C., Smith, B., Patel, S., Coger, T., Dinshaw, A., Patel, S., Huq, S., Musa, M., Rahman, F., Gupta, S., Dolcemascolo, G., & Mann, T. (2021). *Principles for locally led adaptation*.
- Stockholm Environment Institute. (2021). *Effectiveness of climate change adaptation interventions in sub-Saharan Africa and the impact of funding modalities: A mixed methods systematic review protocol.* https://doi.org/10.51414/sei2021.021
- Strauss, A. L. (1987). Qualitative analysis for social scientists. Cambridge university press.
- Strøbech, H. L., & Bordon Rosa, A. (2020). Addressing the risk of maladaptation in Social Protection: The case of World Food Programme. http://lup.lub.lu.se/studentpapers/record/9017585
- Swatuk, L. A., Thomas, B. K., Wirkus, L., Krampe, F., & Batista da Silva, L. P. (2021). The 'boomerang effect': Insights for improved climate action. *Climate and Development*, *13*(1), 61–67. https://doi.org/10.1080/17565529.2020.1723470
- Tabassum, N. (2022). The Politics of Climate Change Knowledge: Labelling Climate Changeinduced Uprooted People (1st ed.). Routledge. https://doi.org/10.4324/9781003038283

- Taylor, M., Eriksen, S., Vincent, K., Brooks, N., Scoville-Simonds, M., & Schipper, L. (2022). Putting'vulnerable groups' at the centre of adaptation interventions by promoting transformative adaptation as a learning process.
- Thomas, K., Hardy, R. D., Lazrus, H., Mendez, M., Orlove, B., Collazo, I., Roberts, J. T., Rockman, M., Warner, B., & Winthrop, R. (2019). Explaining differential vulnerability to climate change: A social science review. *Wiley Interdisciplinary Reviews: Climate Change*, 10. https://doi.org/10.1002/wcc.565
- Thomas, K., & Warner, B. (2019). Weaponizing vulnerability to climate change. *Global Environmental Change*. https://doi.org/10.1016/j.gloenvcha.2019.101928
- Tiwari, A. (2015). The Capacity Crisis in Disaster Risk Management. Springer.
- Torabi, E., Dedekorkut-Howes, A., & Howes, M. (2018). Adapting or maladapting: Building resilience to climate-related disasters in coastal cities. *Cities*, *72*, 295–309. https://doi.org/10.1016/j.cities.2017.09.008
- Tye, S., & Suarez, I. (2021). Locally Led Climate Adaptation: What Is Needed to Accelerate Action and Support? *World Resources Institute*. https://doi.org/10.46830/wriwp.20.00039
- Uitto, J. I., & Batra, G. (Eds.). (2022). *Transformational Change for People and the Planet: Evaluating Environment and Development*. Springer International Publishing. https://doi.org/10.1007/978-3-030-78853-7
- UNDRR. (2022). Vulnerability. https://www.undrr.org/terminology
- UNFCCC. (2007). CLIMATE CHANGE: IMPACTS, VULNERABILITIES AND ADAPTATION IN DEVELOPING COUNTRIES. United Nations Framework Convention on Climate Change.
- Villanueva, P. S., Sword-Daniels, V., Leavy, J., & Wilson, D. (2018). *Tracking and measuring resilience in large programmes: Lessons from BRACED*.
- Webber, S. (2016). Climate Change Adaptation as a Growing Development Priority: Towards Critical Adaptation Scholarship. *Geography Compass*, *10*(10), 401–413. https://doi.org/10.1111/gec3.12278
- Werners, S., Wise, R. M., Butler, J. R. A., Totin, E., & Vincent, K. (2021). Adaptation pathways: A review of approaches and a learning framework. *Environmental Science* & Policy, 116, 266–275. https://doi.org/10.1016/j.envsci.2020.11.003
- Westoby, R., McNamara, K. E., Kumar, R., & Nunn, P. D. (2020). From community-based to locally led adaptation: Evidence from Vanuatu. *Ambio*, *49*(9), 1466–1473. https://doi.org/10.1007/s13280-019-01294-8

- Wilson, R. S., Herziger, A., Hamilton, M., & Brooks, J. S. (2020). From incremental to transformative adaptation in individual responses to climate-exacerbated hazards. *Nature Climate Change*, 10(3), 200–208. https://doi.org/10.1038/s41558-020-0691-6
- Wise, R. M., Fazey, I., Stafford Smith, M., Park, S. E., Eakin, H. C., Archer Van Garderen, E. R.
 M., & Campbell, B. (2014). Reconceptualising adaptation to climate change as part of pathways of change and response. *Global Environmental Change*, 28, 325–336. https://doi.org/10.1016/j.gloenvcha.2013.12.002
- Wojewska, A. N., Singh, C., & Hansen, C. P. (2021). A policy tool for monitoring and evaluation of participation in adaptation projects. *Climate Risk Management*, 33. Scopus. https://doi.org/10.1016/j.crm.2021.100326
- Woodhill, J., & Millican, J. (2023). Systems Thinking and Practice: A guide to concepts, principles and tools for FCDO and partners. Institute of Development Studies. https://doi.org/10.19088/K4D.2023.002
- Work, C., Rong, V., Song, D., & Scheidel, A. (2019). Maladaptation and development as usual? Investigating climate change mitigation and adaptation projects in Cambodia. *Climate Policy*, 19. https://doi.org/10.1080/14693062.2018.1527677
- Yaro, J. A., Teye, J., & Bawakyillenuo, S. (2015). Local institutions and adaptive capacity to climate change/variability in the northern savannah of Ghana. *Climate and Development*, 7(3), 235–245. https://doi.org/10.1080/17565529.2014.951018
- Yin, R. K. (2009). Case study research: Design and methods (Vol. 5). sage.
- Ziervogel, G., Enqvist, J., Metelerkamp, L., & van Breda, J. (2022). Supporting transformative climate adaptation: Community-level capacity building and knowledge co-creation in South Africa. *Climate Policy*, 22(5), 607–622. https://doi.org/10.1080/14693062.2020.1863180

9. Appendices

9.1 Project brief shared beforehand with participants

Title: Analyzing the risk of maladaptation; Implications for planning adaptation projects for development agencies

Climate adaptation has become a pressing concern in the recent decade because of its evident impacts on the social and natural systems despite the efforts in reducing greenhouse gas emissions. While climate adaptation projects around the world are proliferating, emerging theoretical development and empirical case studies provide urgency and a basis to evaluate the consequences of adaptation actions (Eriksen et al., 2021; Simpson et al., 2023). However, the phenomenon and factors that could lead to adaptation becoming a form of maladaptation is still not sufficiently understood or established (Schipper, 2020).

To date, there has been very limited independent review of the positive/negative impacts on social vulnerability of internationally funded adaptation projects. Nevertheless, a large number of theoretical developments and individual empirical case studies are emerging that provide a basis to identify systemic features of the framing, financing, planning, implementation, monitoring and evaluation of adaptation interventions (Atteridge & Remling, 2018; Eriksen et al., 2021; K. Browne & Claudien Razafiarimanana, 2022). Yet, identifying the mechanisms through which negative effects can unfold in adaptation interventions is essential for informing future adaptation policy and actions. Therefore, this study aims to examine how adaptation actions might lead to adverse impacts so that they could be avoided in future planning and implementation of adaptation projects by answering following questions.

- What are the principles that characterize effective adaptation?
- What are the predetermining factors that might lead to negative consequences of adaptation projects?
- What are the challenges in the existing practice for planning and implementing adaptation projects among the I/NGOs? And how these could be addressed?

This study applies a primarily inductive research approach through qualitative descriptive case study, including semi structured interview and project document analysis.

Your participation would be immensely valuable to gain insights on adaptation from practitioners' perspective, and to identify existing challenges and opportunities for effective adaptation.

Thank you!

9.2 Invitation email

Subject: Invitation to participate in a semi-structured interview

Dear [Name],

I hope this email finds you well. I am Bikram Sedhai, and I am a student of Disaster Risk Management and Climate Change Adaptation master's programme at Lund University. I am reaching out to you because I am conducting research on the topic of addressing the risk of maladaptation and I believe that your insights and perspectives would greatly contribute to the findings of my study.

As part of my research, I am seeking participants for a brief interview to gather insights and perspectives on climate adaptation projects. Your participation in this interview would be greatly appreciated and would help contribute to my understanding of the subject.

I would like to invite you to participate in a semi-structured interview, which will take approximately 30 minutes of your time. The purpose of the interview is to gather information and perspectives on the challenges and opportunities in addressing the risk of maladaptation. Your participation in this interview will be significantly valuable in helping to advance our understanding of this important issue. Please find the attached note for the research project.

The interview will be conducted via [preferred method of communication (e.g. Zoom, Skype, Teams)]. If you are interested in participating, please let me know your availability and the preferred method of communication. I will then arrange a convenient time and date for the interview.

Further, I would appreciate you forwarding this email to your colleagues or other potential interviewees you might know.

Thank you for considering this request. I am looking forward to your positive response.

Best regards, Bikram Sedhai <u>Bi2875se-s@student.lu.se</u>

9.3 Interview consent form

Thesis Title: Analyzing the risk of maladaptation; Implications for planning adaptation projects for development agencies

Purpose of the study: This study aims to examine how adaptation actions might lead to adverse impacts so that they could be avoided in future adaptation projects among international non-governmental organizations. Overall objective of the research project is to identify potential ways of addressing the risk of maladaptation and providing inputs on planning and implementing adaptation projects for development agencies.

Procedure: You have been asked to participate in a semi-structured interview as part of this study. The interview will last approximately 30 minutes and will take place online. The interview will be recorded and transcribed for the purposes of analysis. The main purpose of this interview is to understand and explore what adaptation means and different associated factors from practitioners' perspective. Interview also aims to identify existing challenges and opportunities in adaptation practice and potential recommendations to reduce the risk of maladaptation.

- I understand that my participation in this study is completely voluntary. I may choose not to participate, or withdraw from the study at any time of the study without any consequences.
- I understand that I can stop the interview at any time if I feel uncomfortable.
- Please Choose one of the following:

I understand that I can choose to be quoted and referenced and I have no need for and waiver my right for anonymity.
I understand that I can choose to remain anonymous regarding any references to what I say in the interviews. No quotes will be attributed to me and any information I give will be paraphrased in a way that cannot be traced back to myself.

- I understand that all data, including personal data, will be treated with utmost confidentiality and according to GDPR rules and Lund University's research code of conduct
- I understand that I am free to contact the researcher at any time to seek further clarification and information.

By signing below, I indicate that I have read and understood the above information and that I voluntarily agree to participate in this study.

Name of participant:

Date:

Signature:

9.4 Generic interview guide

Following interview guide is intended to support the data collection through a semi structured interview. Depending on the background and context of the interviewees, the questions could be used partly or fully, and adapted as suited, and the order of the questions could also be adapted accordingly.

Segment	Interview question	Probing question
General Introduction	 Can you please introduce yourself and your area of work? Can you tell us a bit about your background? Can you provide an overview of your organization's mission and how it relates to climate adaptation? 	
Segment I	Open ended questions to establish a mutual understanding of key concepts and letting interviewee have their own narrative. (Input to research question 1 and 2)	4. Could you please elaborate a bit more?
	4. In your opinion, how is climate change adaptation understood by your organization?5. What is effective adaptation for you and your work?	organization consider as maladaptive?
	 6. Are you familiar with the concept of maladaptation²? 7. As a practitioner, do you think there are benefits of applying such concepts in adaptation projects? 	b. If no , share the common definition of maladaptation. What do you think of this concept? Do you think it is a useful idea for INGOs?
Segment II	 Input to research questions 1 and 2 (focused and detailed questions) Adaptation, as a field of study has identified different dimensions of adaptation initiatives and presents that not all adaptation reduces vulnerability. Not all adaptation will 'do good'; there will be trade-offs, feedbacks and negative consequences. 8. how does your organization address the potential 	8. How do you characterize whether a project is effective or not?Do you think adaptation projects could somehow exacerbate the issue that are sought to be solved?
	 9. How does your organization address the potential unintended consequences in the context of climate change adaptation projects? 9. How does your organization assess/measure the success of adaptation interventions? 10. In your experience, what are the factors that influence the effectiveness of your action? 	11 a. If yes, how were these outcomes identified and addressed? Were there any warnings, signs, or indicators that these maladaptive outcomes could occur?

² adaptation measures that do not succeed in reducing vulnerability but increase it instead. Briefly mention about unintended consequences of projects

	11. Have you observed any examples of negative consequences of adaptation projects in the communities you work with?12. To what extent do you think adaptation projects acknowledge contextual vulnerability of the communities you work with?	b. If no, how does the projects ensure that potential negative consequences are considered and addressed during planning and implementation?
	 For only interviewees involved in three case study projects 13. How does the xxx project fit into a larger, long-term adaptation need/strategy for the community or region? 14. What is the xxx project main aim? 15. In your opinion, what were the three key successes? And three key learnings? 16. After the completion of project, what do you see as a major risk with xxx projects? 17. Looking back to the completed project, have you noticed any project intervention that produced counter-productive effects in relation to vulnerability or potential impacts that were formally intended to be addressed? 18. How and to what extent do you think the framing of adaptation? Who should adapt? What and how to adapt? 19. What was the opportunity for the project to facilitate adaptation? 20. The xxx project document mentioned some potential socio-environmental impacts of adaptation project, how were those considered/mitigated during project implementation phase? 	 14. a. How do you think the project addressed the dynamic vulnerability? b. In your opinion, how are current planning, implementation, and MEL instrument effective to do this? 15. What would you like to change if you get to plan and implement a similar project again? 16.a Do you think maladaptation could be a risk to a project? b. Is there any other tool or mechanism like risk matrix which captures this? 19. Opportunity could be any tools, policy, learning, innovation, etc.
Segment III	Input to research question 3	,
	 21. Can you describe your organization's approach to climate adaptation? What specific measures have you implemented to address the impacts of climate change via adaptation projects? 22. What challenges has your organization faced in implementing climate adaptation measures? 23. Could you please elaborate on how maladaptation is currently conceptualized or considered during different phases of the projects? 	24 a. How participatory are such processes?b. How are the temporal and spatial boundaries of evaluation determined?

	 24. Could you provide a brief overview of the evaluation process of the adaptation projects? 25. What are some of the key lessons that your organization has learned in terms of addressing climate change and promoting adaptation? 26. In your opinion what can be done/improved concerning the adaptation outcomes? Any processes, policies, stakeholders? 	c. In the current practice, how is maladaptation incorporated during project evaluations?
Ending	 Return to any points that were brought up previously in need of further clarification. Allow interviewee to bring up any points not discussed. 27. What recommendations would you have for future climate adaptation projects to avoid maladaptive outcomes? 28. Do you have any questions or comments you would like to add before ending the interview? 29. Are there any topics you find important in this context that we have not discussed today? 30. Are there any other people you think I should talk to? 	

For Academics and others

- Can you please introduce yourself and your area of work?
- After research, how was the response from international development agencies, donor organizations? Have you continued researching on this topic?
- How is adaptation framed in the current mainstream research? And how do you think it has shaped the adaptation practice in the international arena/internationally funded projects?
- Ho do you define a successful adaptation? Any characteristics or principles of effective adaptation? How should development organizations measure/evaluate effects of adaptation projects?
- What do you think are the main conceptual strength and limitations of framing and defining maladaptation?
- Maladaptation as articulated in the literature and your research takes many forms, in your opinion what do you think are the factors that determine such pathways?

- Where are the causes of maladaptation located? Do these lie outside the realm of adaptation?
- Could maladaptive outcomes been reasonably expected or foreseen by practitioners? How should it be accounted in different phases of project management?
- What challenges do you see in operationalization of maladaptation in adaptation projects? Could you elaborate on potential opportunities for considering maladaptation risk while planning?
- In the context of multiple adaptation pathways, how do you think INGOs could see beyond climate impacts? Because they need
- What do you think about the current bilateral/multilateral funding mechanism or overall practice of development agencies? Do they address the potential risk of maladaptation? What do you think could be improved?
- What do you think of subjective and value laden nature of adaptation/maladaptation? And what does this look like in practice? How is subjectivity to be addressed in adaptation assessments?
- Would you like to give any suggestions for adaptation practitioners to develop projects with low risk of maladaptive outcome?
- Do you have any questions or comments you would like to add before ending the interview?
- Are there any topics you find important in this context that we have not discussed today?
- Would you recommend any research project that I should refer?